

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

November 4, 2016

Marc S. Gerber Skadden, Arps, Slate, Meagher & Flom LLP mgerber@skadden.com

Re: Oshkosh Corporation

Incoming letter dated September 20, 2016

Dear Mr. Gerber:

This is in response to your letter dated September 20, 2016 concerning the shareholder proposal submitted to Oshkosh by John Chevedden. Copies of all of the correspondence on which this response is based will be made available on our website at http://www.sec.gov/divisions/corpfin/cf-noaction/14a-8.shtml. For your reference, a brief discussion of the Division's informal procedures regarding shareholder proposals is also available at the same website address.

Sincerely,

Matt S. McNair Senior Special Counsel

Enclosure

cc: John Chevedden

FISMA & OMB Memorandum M-07-16

Response of the Office of Chief Counsel Division of Corporation Finance

Re: Oshkosh Corporation

Incoming letter dated September 20, 2016

The proposal asks the board to amend certain provisions of the company's proxy access bylaw in the manner specified in the proposal.

There appears to be some basis for your view that Oshkosh may exclude the proposal under rule 14a-8(i)(10). Based on the information you have presented, it appears that Oshkosh's policies, practices and procedures compare favorably with the guidelines of the proposal and that Oshkosh has, therefore, substantially implemented the proposal. Accordingly, we will not recommend enforcement action to the Commission if Oshkosh omits the proposal from its proxy materials in reliance on rule 14a-8(i)(10).

Sincerely,

Evan S. Jacobson Special Counsel

DIVISION OF CORPORATION FINANCE INFORMAL PROCEDURES REGARDING SHAREHOLDER PROPOSALS

The Division of Corporation Finance believes that its responsibility with respect to matters arising under Rule 14a-8 [17 CFR 240.14a-8], as with other matters under the proxy rules, is to aid those who must comply with the rule by offering informal advice and suggestions and to determine, initially, whether or not it may be appropriate in a particular matter to recommend enforcement action to the Commission. In connection with a shareholder proposal under Rule 14a-8, the Division's staff considers the information furnished to it by the company in support of its intention to exclude the proposal from the company's proxy materials, as well as any information furnished by the proponent or the proponent's representative.

Although Rule 14a-8(k) does not require any communications from shareholders to the Commission's staff, the staff will always consider information concerning alleged violations of the statutes and rules administered by the Commission, including arguments as to whether or not activities proposed to be taken would violate the statute or rule involved. The receipt by the staff of such information, however, should not be construed as changing the staff's informal procedures and proxy review into a formal or adversarial procedure.

It is important to note that the staff's no-action responses to Rule 14a-8(j) submissions reflect only informal views. The determinations reached in these no-action letters do not and cannot adjudicate the merits of a company's position with respect to the proposal. Only a court such as a U.S. District Court can decide whether a company is obligated to include shareholder proposals in its proxy materials. Accordingly, a discretionary determination not to recommend or take Commission enforcement action does not preclude a proponent, or any shareholder of a company, from pursuing any rights he or she may have against the company in court, should the company's management omit the proposal from the company's proxy materials.

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP

1440 NEW YORK AVENUE, N.W. WASHINGTON, D.C. 20005-2111

TEL: (202) 371-7000 FAX: (202) 393-5760 www.skadden.com

DIRECT DIAL
202-37 I -7233
DIRECT FAX
202-66 I -8280
EMAIL ADDRESS
MGERBER@SKADDEN.COM

BY EMAIL (shareholderproposals@sec.gov)

FIRM/AFFILIATE OFFICES BOSTON CHICAGO HOUSTON LOS ANGELES NEW YORK PALO ALTO WILMINGTON BEIJING BRUSSELS FRANKFURT LONDON MOSCOW MUNICH PARIS SÃO PAULO SEOUL SHANGHAI SINGAPORE TOKYO TORONTO

September 20, 2016

U.S. Securities and Exchange Commission Division of Corporation Finance Office of Chief Counsel 100 F Street, N.E. Washington, D.C. 20549

> RE: Oshkosh Corporation – 2017 Annual Meeting Omission of Shareholder Proposal of John Chevedden

Ladies and Gentlemen:

Pursuant to Rule 14a-8(j) promulgated under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), we are writing on behalf of our client, Oshkosh Corporation, a Wisconsin corporation (the "Company"), to request that the Staff of the Division of Corporation Finance (the "Staff") of the U.S. Securities and Exchange Commission (the "Commission") concur with the Company's view that, for the reasons stated below, it may exclude the shareholder proposal and supporting statement (the "Proposal") submitted by John Chevedden (the "Proponent") from the proxy materials to be distributed by the Company in connection with its 2017 annual meeting of shareholders (the "2017 proxy materials").

In accordance with Section C of Staff Legal Bulletin No. 14D (Nov. 7, 2008) ("SLB 14D"), we are emailing this letter and its attachments to the Staff at shareholderproposals@sec.gov. In accordance with Rule 14a-8(j), we are simultaneously sending a copy of this letter and its attachments to the Proponent as notice of the Company's intent to omit the Proposal from the 2017 proxy materials.

Rule 14a-8(k) and Section E of SLB 14D provide that shareholder proponents are required to send companies a copy of any correspondence that the shareholder proponents elect to submit to the Commission or the Staff. Accordingly, we are taking this opportunity to remind the Proponent that if the Proponent submits correspondence to the Commission or the Staff with respect to the Proposal, a copy of that correspondence should concurrently be furnished to the Company.

I. The Proposal

The resolution of the Proposal is copied below:

RESOLVED: Shareholders ask our Board of Directors to adopt, and present for shareholder approval, an enhancement package for the company bylaws allowing shareholder nominated candidates to be included in the company's proxy materials, with essential elements for substantial implementation as follows:

- 1. The "Minimum Number" of shares of the Company's common stock required to nominate shall be 3%, instead of 5%, as specified in current bylaws.
- 2. The number of shareholder-nominated candidates eligible to appear in proxy materials shall be one quarter of the directors then serving or two, whichever is greater. With twelve directors, current bylaws allow only up to two proxy access candidates. These amendments will allow for up to three, potentially enabling proxy access candidates to serve on each of the Company's three committees without being over-extended.
- 3. No limitation shall be placed on the number of shareholders that can aggregate their shares to achieve the 3% "Required Shares," outstanding shares of the Company entitled to vote in the election of directors. Under current provisions, even if the 20 largest public pension funds were able to aggregate their shares, they would not meet the 3% criteria at most companies examined by the Council of Institutional Investors.
- 4. No limitation shall be placed on the re-nomination of shareholder nominees based on the number or percentage of votes received in any election. Such limitations do not facilitate the shareholders' traditional state law rights and add unnecessary complexity.

- 5. The bylaws shall not require that a nominator provide a statement of intent to continue to hold the required percentage of shares after the annual meeting.
- 6. Loaned securities shall be counted as belonging to a nominating shareholder if the shareholder represents it:
 - (a) has the legal right to recall those securities for voting purposes,
 - (b) will vote the securities at the shareholder meeting and
 - (c) will hold those securities through the date of the annual meeting.

Loaning securities to a third party with recall provisions greater than five days is not inconsistent with a long-term investment in a company.

II. Basis for Exclusion

We hereby respectfully request that the Staff concur in the Company's view that it may exclude the Proposal from the 2017 proxy materials pursuant to Rule 14a-8(i)(10) because the Company has substantially implemented the Proposal.

III. Background

A. The Proposal

The Company received the Proposal on August 8, 2016, accompanied by a cover letter from the Proponent. Copies of the Proposal and cover letter are attached hereto as Exhibit A.

B. The Company's Recent By-Law Amendment

As described in greater detail below, on September 13, 2016, the Board of Directors (the "Board") amended the Company's By-Laws (as amended, the "Amended By-Laws") to reduce the "Minimum Number" of shares of the Company's common stock an "Eligible Holder" must "own" to make a proxy access nomination from 5% to 3% and to eliminate certain other provisions of the previous

proxy access by-law. The Amended By-Laws, which were effective immediately, are included as an exhibit to the Company's Current Report on Form 8-K, filed with Commission on September 16, 2016, and that Form 8-K is attached hereto as Exhibit B.

IV. The Proposal May be Excluded Under Rule 14a-8(i)(10) Because the Company Has Substantially Implemented the Proposal

A. Rule 14a-8(i)(10)

Rule 14a-8(i)(10) permits a company to exclude a shareholder proposal if the company has already substantially implemented the proposal. The Commission adopted the "substantially implemented" standard in 1983 after determining that the "previous formalistic application" of the rule defeated its purpose, which is to "avoid the possibility of shareholders having to consider matters which already have been favorably acted upon by the management." *See* Exchange Act Release No. 34-20091 (Aug. 16, 1983) (the "1983 Release") and Exchange Act Release No. 12598 (July 7, 1976). Accordingly, the actions requested by a proposal need not be "fully effected" provided that they have been "substantially implemented" by the company. *See* 1983 Release.

Applying this standard, the Staff has permitted exclusion under Rule 14a-8(i)(10) when the company's policies, practices and procedures compare favorably with the guidelines of the proposal. See, e.g., Exxon Mobil Corp. (Mar. 17, 2015) (permitting exclusion of a proposal requesting that the company commit to increasing the dollar amount authorized for capital distributions to shareholders through dividends or share buybacks where the company's long-standing capital allocation strategy and related "policies practices and procedures compare[d] favorably with the guidelines of the proposal and...therefore, substantially implemented the proposal"); Walgreen Co. (Sept. 26, 2013) (permitting exclusion of a proposal requesting elimination of certain supermajority vote requirements where the company's elimination from its governing documents of all but one such requirement "compare[d] favorably with the guidelines of the proposal"); General Dynamics Corp. (Feb. 6, 2009) (permitting exclusion of a proposal requesting a 10% ownership threshold for special meetings where the company planned to adopt a special meeting bylaw with an ownership threshold of 10% for special meetings called by one shareholder and 25% for special meetings called by a group of shareholders).

We believe that, with the Company's change to a 3% proxy access bylaw, there are only five companies remaining that have proxy access bylaws requiring 5% minimum ownership for three years in order to nominate up to 20% or 25% of the board.

In addition, the Staff has permitted exclusion under Rule 14a-8(i)(10), even if the proposal has not been implemented exactly as proposed by the shareholder proponent, where a company has satisfied the essential objective of the proposal. See, e.g., AGL Resources Inc. (granted on recon., Mar. 5, 2015) (permitting exclusion of a proposal seeking to grant holders of 25% of the company's outstanding shares the power to call a special meeting where the board approved, and undertook to submit for shareholder approval, an amendment to the articles of incorporation to grant shareholders holding for at least one year 25% of the outstanding shares the power to call a special meeting); Textron, Inc. (Jan. 21, 2010) (permitting exclusion of a proposal requesting immediate board declassification where the board submitted a phased-in declassification proposal for shareholder approval); Hewlett-Packard Co. (Dec. 11, 2007) (permitting exclusion of a proposal requesting the ability for shareholders to call special meetings where the board had proposed a bylaw amendment allowing shareholders to call a special meeting unless the business to be proposed at that meeting recently had been, or soon would be, addressed at an annual meeting).

Applying these principles, the Staff has permitted exclusion under Rule 14a-8(i)(10) of a proposal to adopt a proxy access bylaw with a 3% minimum ownership requirement where the company adopted a proxy access bylaw with that minimum ownership requirement and thereby "addresse[d] the proposal's essential objective." In these instances, the Staff agreed that the proposal had been substantially implemented even though the proposal sought a proxy access bylaw with no limits on the ability of shareholders to aggregate their shares to satisfy the minimum ownership requirement or sought to limit proxy access nominees to 25% of the total number of directors and the bylaw adopted by the company contained a 20-person aggregation limit or limited proxy access nominees to 20% of the total number of directors. See, e.g., Oracle Corp. (Aug. 11, 2016); Cardinal Health, Inc. (July 20, 2016); Leidos Holdings, Inc. (May 4, 2016); Equinix, Inc. (Apr. 7, 2016); Amphenol Corp. (granted on recon., Mar. 29, 2016); Omnicom Group Inc. (Mar. 22, 2016); General Motors Co. (Mar. 21, 2016); Quest Diagnostics Inc. (Mar. 17, 2016); Chemed Corp. (Mar. 9, 2016); Eastman Chemical Co. (Mar. 9, 2016); Newell Rubbermaid Inc. (Mar. 9, 2016); Amazon.com, Inc. (Mar. 3, 2016); Anthem, Inc. (Mar. 3, 2016); Fluor Corp. (Mar. 3, 2016); International Paper Co. (Mar. 3, 2016); ITT Corp. (Mar. 3, 2016); McGraw Hill Financial, Inc. (Mar. 3, 2016); PG&E Corp. (March 3, 2016); Public Service Enterprise Group Inc. (Mar. 3, 2016); Sempra Energy (Mar. 3, 2016); Xylem Inc. (Mar. 3, 2016); The Wendy's Co. (Mar. 2, 2016); Reliance Steel & Aluminum Co. (Feb. 26, 2016); United Continental Holdings, Inc. (Feb. 26, 2016); Alaska Air Group, Inc. (Feb. 12, 2016); Baxter Int'l Inc. (Feb. 12, 2016); Capital One Financial Corp. (Feb. 12, 2016); The Dun & Bradstreet Corp. (Feb. 12, 2016); General Dynamics Corp. (Feb. 12, 2016); Huntington Ingalls Industries, Inc. (Feb. 12, 2016); Illinois Tool Works, Inc. (Feb. 12, 2016); Northrop

Grumman Corp. (Feb. 12, 2016); PPG Industries, Inc. (Feb. 12, 2016); Science Applications Int'l Corp. (Feb. 12, 2016); Target Corp. (Feb. 12, 2016); Time Warner, Inc. (Feb. 12, 2016); UnitedHealth Group, Inc. (Feb. 12, 2016); The Western Union Co. (Feb. 12, 2016).

B. The Recent Amendment to the Company's By-Laws Satisfies the Proposal's Essential Objective

As referenced above, on September 13, 2016, the Board amended the Company's By-Laws to reduce the "Minimum Number" of shares of common stock an "Eligible Holder" must "own" to make a proxy access nomination from 5% to 3%. In addition, the previous proxy access by-law was amended to eliminate the requirement that a proxy access nominee receive at least 25% of the votes cast at an annual meeting to be eligible for re-nomination at one of the next two annual meetings and to eliminate the requirement to make a representation as to the shareholder's intentions with respect to continuing to own the "Minimum Number" of shares of Company common stock for at least one year following the annual meeting (as so amended, the "Amended Proxy Access Provision").

Consistent with the precedent described above, the Amended Proxy Access Provision satisfies the Proposal's essential objective – expanding the ability of shareholders to use proxy access, including by providing a shareholder or group of shareholders that have owned 3% or more (rather than 5% or more) of the Company's common stock continuously for at least three years the ability to include no fewer than two director nominees in the Company's annual meeting proxy materials.

Moreover, in *NVR*, *Inc.* (*granted on recon.*, Mar. 25, 2016) the Staff was presented with substantially similar circumstances and concurred with the company's view that its amendment to its proxy access bylaw, including reducing the 5% minimum ownership requirement to 3%, "compare[d] favorably with the guidelines of the proposal." Specifically, in *NVR*, the company received a shareholder proposal seeking to amend a previously-adopted 5% proxy access bylaw in the following four respects:

- To reduce the number of shares of common stock an "Eligible Shareholder" must "own" from 5% to 3% of the class outstanding;
- To eliminate the 20-person limit on the number of shareholders who may aggregate their shares to meet the minimum ownership requirement (and the related provision as to how to count related funds);

- To provide that a shareholder will be deemed to "own" shares loaned to another person if the shareholder may recall the shares within five business days (as opposed to three business days); and
- To revise the requirement that a nominating shareholder represent that it intends to own the "Required Shares" for at least one year following the date of the annual meeting to require that the shareholder represent only that it will remain a shareholder during the one-year period.

Following receipt of the proposal, NVR amended its proxy access bylaw to reduce the minimum ownership requirement from 5% to 3% of the outstanding common stock and to increase the permissible recall period for loaned shares from three business days to five business days. NVR did not eliminate the 20-person limit on the number of shareholders who may aggregate their shares to meet the 3% minimum ownership requirement and did not revise the requirement that a nominating shareholder represent that it intends to own the minimum number of shares for at least one year following the annual meeting. The Staff agreed that NVR had substantially implemented the proposal.

Similarly, in the instant case, the Company's September adoption of the Amended Proxy Access Provision compares favorably with the guidelines of the Proposal such that the Proposal has been substantially implemented. Specifically, the Proposal requests that the prior version of the Company's proxy access by-law be amended in the following six respects:

- To reduce the "Minimum Number" of shares of the Company's common stock an "Eligible Holder" must "own" from 5% to 3%;
- To increase the maximum number of proxy access nominees that must be included in the Company's proxy materials from 20% of the total number of directors (but not less than two) to 25% of the total number of directors (but not less than two);
- To eliminate the 20-person limit on the number of shareholders who may aggregate their shares to meet the minimum ownership requirement;
- To eliminate the requirement that a proxy access nominee receive at least 25% of the votes cast at an annual meeting to be eligible for renomination at one of the next two annual meetings;

- To eliminate the requirement that the Nominating Shareholder provide a representation as to its intentions with respect to continuing to own the "Minimum Number" of shares of Company common stock for at least one year following the annual meeting; and
- To count loaned securities as "owned" so long as a shareholder makes certain representations, without a requirement that the shareholder have the power to recall such loaned shares on five business days' notice.

As described above, following receipt of the Proposal, the Company has amended the prior version of its proxy access by-law to reduce the minimum ownership requirement from 5% to 3% of the Company's common stock, to eliminate the requirement that a proxy access nominee receive at least 25% of the votes cast at an annual meeting to be eligible for re-nomination and to eliminate the requirement to provide a representation regarding intentions with respect to continuing to own the minimum number of shares of Company common stock for at least one year following the annual meeting. Although the Proposal seeks to amend the prior proxy access by-law in six respects, the Company has now adopted three of the proposed amendments, including the 3% minimum ownership requirement, which has generally been considered to be one of the most "essential" elements of proxy access. The Amended Proxy Access Provision significantly expands the ability of shareholders to use proxy access and thereby satisfies the Proposal's essential objective.

The remaining amendments proposed by the Proponent relate to the maximum number of access nominees (a limit of 25% of the number of directors versus 20%, but with a minimum of two nominees under either limit), the 20-person aggregation limit to meet the minimum ownership requirement and certain details of when to include loaned shares in the amount of shares considered "owned." As described above, the Staff has previously agreed that amending a 5% proxy access bylaw to a 3% proxy access bylaw while retaining a 20-person aggregation limit substantially implements a proposal seeking to amend a 5% proxy access bylaw to a 3% proxy access bylaw and to remove aggregation limits (among other things). See NVR, Inc. (granted on recon., Mar. 25, 2016). Similarly, the Staff has agreed that adoption of a 3% proxy access bylaw permitting nominees for up to 20% of the number of directors substantially implements a proposal seeking 3% proxy access for up to 25% of the number of directors. See, e.g., Oracle Corp. (Aug. 11, 2016); Cardinal Health, Inc. (July 20, 2016); Leidos Holdings, Inc. (May 4, 2016); Equinix, Inc. (Apr. 7, 2016). Finally, the Company's proxy access by-law already provided, and continues to provide, that loaned shares count as "owned" for proxy access purposes, subject to certain requirements. The existing five business day recall

requirement is not a practical impediment to a "useable" proxy access right and, in the context of this particular Proposal and the Amended Proxy Access Provision, does not nullify the amendments' satisfaction of the Proposal's essential objective.

We note that the Staff has considered and rejected substantial implementation arguments in two other instances concerning proposals to amend existing proxy access rights, both of which are inapposite to the facts at issue here. In SBA Communications Corp. (Feb. 12, 2016), the proposal, similar to the Proposal, sought to amend a 5% proxy access bylaw to reduce the minimum ownership requirement to 3% and make certain other changes. Unlike the Company, however, SBA Communications did not amend its proxy access bylaw in response to the proposal and argued that its previous adoption of a 5% proxy access bylaw substantially implemented the proposal seeking to amend the bylaw to reduce the minimum ownership requirement to 3%. In H&R Block, Inc. (July 21, 2016), the proposal sought certain amendments to a previously-adopted proxy access bylaw but the minimum ownership requirement was not part of the proposal as the company had already adopted a 3% minimum ownership requirement. Like SBA Communications, H&R Block did not amend its proxy access bylaw in response to the proposal and argued that its previous adoption of a proxy access bylaw substantially implemented the proposal seeking to amend the bylaw in various respects. Whereas the Company in this instance did amend its proxy access by-law in response to the Proposal, neither of the no-action requests in SBA Communications or *H&R Block* involved an analysis of whether amendments to an existing proxy access bylaw adopted in response to the receipt of a proposal seeking amendments to the proxy access bylaw substantially implemented that proposal.

As in *NVR*, *Inc.*, the Company's Amended Proxy Access Provision satisfies the Proposal's essential objective – expanding the ability of shareholders to use proxy access, including by providing a shareholder or group of shareholders that have owned 3% or more (rather than 5% or more) of the Company's common stock continuously for at least three years the ability to include no fewer than two director nominees in the Company's annual meeting proxy materials. Thus, even though the Amended Proxy Access Provision has not been implemented exactly as proposed by the Proponent, the Company has substantially implemented the Proposal. Accordingly, the Company believes the Proposal is excludable under Rule 14a-8(i)(10).

V. Conclusion

Based upon the foregoing analysis, the Company respectfully requests that the Staff concur that it will take no action if the Company excludes the Proposal from its 2017 proxy materials. Should the Staff disagree with the conclusions set

forth in this letter, or should any additional information be desired in support of the Company's position, we would appreciate the opportunity to confer with the Staff concerning these matters prior to the issuance of the Staff's response. Please do not hesitate to contact the undersigned at (202) 371-7233 or Richard J. Grossman at (212) 735-2116.

Very truly yours

Marc S. Gerber

cc: John Chevedden

EXHIBIT A

(see attached)

JOHN CHEVEDDEN

*** FISMA & OMB Memorandum M-07-16 ***

*** FISMA & OMB Memorandum M-07-16 ***

Mr. Ignacio Cortina Corporate Secretary Oshkosh Corporation (OSK) 2307 Oregon Street, P.O. Box 2566 Oshkosh, WI 54903-2566 PH: 920-233-9301

PH: 920-233-9301 FX: 920-237-4228

Dear Mr. Cortina,

This Rule 14a-8 proposal is respectfully submitted in support of the long-term performance of our company. This Rule 14a-8 proposal is intended as a low-cost method to improve compnay performance. This proposal is for the next annual shareholder meeting. Rule 14a-8 requirements will be met including the continuous ownership of the required stock value until after the date of the respective shareholder meeting and presentation of the proposal at the annual meeting. This submitted format, with the shareholder-supplied emphasis, is intended to be used for definitive proxy publication.

Your consideration and the consideration of the Board of Directors is appreciated in support of the long-term performance of our company. Please acknowledge receipt of this proposal by email to FISMA & OMB Memorandum M-07-16 ***

August 8, 2016

Sincerely.

Khn Chevedden

cc: Lynn Liebergen < lliebergen @oshkoshcorp.com>

Pamela Patzke <ppatzke@oshkoshcorp.com>

[OSK – Rule 14a-8 Proposal, August 8, 2016 – This line is not for publication] Proposal [4] - Shareholder Proxy Access Enhancement

RESOLVED: Shareholders ask our Board of Directors to adopt, and present for shareholder approval, an enhancement package for the company bylaws allowing shareholder nominated candidates to be included in the company's proxy materials, with essential elements for substantial implementation as follows:

- 1. The "Minimum Number" of shares of the Company's common stock required to nominate shall be 3%, instead of 5%, as specified in current bylaws.
- 2. The number of shareholder-nominated candidates eligible to appear in proxy materials shall be one quarter of the directors then serving or two, whichever is greater. With twelve directors, current bylaws allow only up to two proxy access candidates. These amendments will allow for up to three, potentially enabling proxy access candidates to serve on each of the Company's three committees without being over-extended.
- 3. No limitation shall be placed on the number of shareholders that can aggregate their shares to achieve the 3% "Required Shares," outstanding shares of the Company entitled to vote in the election of directors. Under current provisions, even if the 20 largest public pension funds were able to aggregate their shares, they would not meet the 3% criteria at most companies examined by the Council of Institutional Investors.
- 4. No limitation shall be placed on the re-nomination of shareholder nominees based on the number or percentage of votes received in any election. Such limitations do not facilitate the shareholders' traditional state law rights and add unnecessary complexity.
- 5. The bylaws shall not require that a nominator provide a statement of intent to continue to hold the required percentage of shares after the annual meeting.
- 6. Loaned securities shall be counted as belonging to a nominating shareholder if the shareholder represents it:
- (a) has the legal right to recall those securities for voting purposes,
- (b) will vote the securities at the shareholder meeting and
- (c) will hold those securities through the date of the annual meeting.

Loaning securities to a third party with recall provisions greater than five days is not inconsistent with a long-term investment in a company.

Shareholder proxy access at US companies would "benefit both the markets and corporate boardrooms, with little cost or disruption," raising US market capitalization by up to \$140 billion. This is according to a cost-benefit analysis by the Chartered Financial Analyst Institute, *Proxy Access in the United States: Revisiting the Proposed SEC Rule.* Although our company adopted a proxy access bylaw, it contains troublesome provisions that significantly impair the ability of shareholders to use it. Adoption of this requested enhancement package would largely remedy that situation.

Please vote to enhance shareholder value:

Shareholder Proxy Access Enhancement – Proposal [4]

[The above line is for publication.]

sponsors this

Notes:

This proposal is believed to conform with Staff Legal Bulletin No. 14B (CF), September 15, 2004 including (emphasis added):

Accordingly, going forward, we believe that it would not be appropriate for companies to exclude supporting statement language and/or an entire proposal in reliance on rule 14a-8(I)(3) in the following circumstances:

- the company objects to factual assertions because they are not supported;
- the company objects to factual assertions that, while not materially false or misleading, may be disputed or countered;
- the company objects to factual assertions because those assertions may be interpreted by shareholders in a manner that is unfavorable to the company, its directors, or its officers; and/or
- the company objects to statements because they represent the opinion of the shareholder proponent or a referenced source, but the statements are not identified specifically as such.

We believe that it is appropriate under rule 14a-8 for companies to address these objections in their statements of opposition.

See also: Sun Microsystems, Inc. (July 21, 2005).

The stock supporting this proposal will be held until after the annual meeting and the proposal will be presented at the annual meeting. Please acknowledge this proposal promptly by email

^{***} FISMA & OMB Memorandum M-07-16 ***

EXHIBIT B

(see attached)

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

			
	F	ORM 8-K	
	CURR	ENT REPORT	
		Section 13 or 15(d) of Exchange Act of 1934	
	Date of Report (Date of earliest event reported):	September 13, 201	16
		H CORPORATION rant as specified in its cha	arter)
Wisconsin (State or other jurisdiction of incorporation)	(Cor	1-31371 mmission File Number)	39-0520270 (IRS Employer Identification No.)
(Oshkosh, Wisconsin 54903 cutive offices, including z	
		0) 235-9151 e number, including area o	code)
Check the appropriate box below if t under any of the following provisions:	he Form 8-K filing is in	ntended to simultaneously	satisfy the filing obligation of the registrant
[] Written communications pursuant	to Rule 425 under the Se	ecurities Act (17 CFR 230	.425)
[] Soliciting material pursuant to Rule	14a-12 under the Excha	ange Act (17 CFR 240.14	a-12)
[] Pre-commencement communication	ns pursuant to Rule 14d	-2(b) under the Exchange	Act (17 CFR 240.14d-2(b))
[] Pre-commencement communication	ons pursuant to Rule 13e-	-4(c) under the Exchange	Act (17 CFR 240.13e-4(c))

<u>Item 5.03</u> <u>Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.</u>

In November 2015, the board of directors (the "Board") of Oshkosh Corporation (the "Company") amended the Company's By-Laws to implement "proxy access" (allowing eligible shareholders to include their own nominees for director in the Company's proxy materials along with Board-nominated nominees). On September 13, 2016, the Board amended the Company's By-Laws (as so amended, the "Amended By-Laws") to amend certain terms of the proxy access provisions of the Company's By-Laws (the "Proxy Access Provisions").

The amendments to the Proxy Access Provisions effected the following three changes:

- reduced the ownership threshold from 5% to 3% for any shareholder or group of up to 20 shareholders who have
 maintained continuous qualifying ownership of at least that amount of the Company's outstanding common stock for at
 least the previous three years to include a specified number of director nominees in the Company's proxy materials for
 an annual meeting of shareholders;
- 2) eliminated the requirement that a nominating shareholder make a representation and warranty to the Company with respect to the nominating shareholder's intentions to maintain qualifying ownership of shares of the Company's outstanding common stock for at least one year after the annual meeting of shareholders at which the nominating shareholder intends to nominate directors for election; and
- 3) removed the right of the Company to omit from its proxy materials a shareholder nominee nominated under the Proxy Access Provisions if the nominee was nominated pursuant to the Proxy Access Provisions at one of the Company's two preceding annual meetings and received a vote of less than 25% of the shares entitled to vote.

The foregoing general description of the amendments to the Proxy Access Provisions is qualified in its entirety by reference to the full text of the Amended By-Laws, a copy of which is filed as Exhibit 3.1 to this report and incorporated by reference herein.

Item 9.01. Financial Statements and Exhibits.

- (a) Not applicable.
- (b) Not applicable.
- (c) Not applicable.
- (d) Exhibits. The following exhibit is being filed herewith:
 - (3.1) By-Laws of Oshkosh Corporation, as amended effective September 13, 2016.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

OSHKOSH CORPORATION

By: <u>/s/ Ignacio A. Cortina</u> Ignacio A. Cortina Date: September 16, 2016

Senior Vice President, General Counsel

and Secretary

OSHKOSH CORPORATION

Exhibit Index to Current Report on Form 8-K Dated September 13, 2016

Exhibit Number

(3.1) By-Laws of Oshkosh Corporation, as amended effective September 13, 2016.

BY-LAWS

OF

OSHKOSH CORPORATION (a Wisconsin corporation)

ARTICLE I. OFFICES

- 1.01 <u>Principal and Business Offices</u>. The corporation may have such principal and other business offices, either within or without the State of Wisconsin, as the Board of Directors may designate or as the business of the corporation may require from time to time.
- 1.02 <u>Registered Office</u>. The registered office of the corporation required by the Wisconsin Business Corporation Law to be maintained in the State of Wisconsin may be, but need not be, identical with the principal office in the State of Wisconsin, and the address of the registered office may be changed from time to time by the Board of Directors or by the registered agent. The business office of the registered agent of the corporation shall be identical to such registered office.

ARTICLE II. SHAREHOLDERS

2.01 Annual Meeting. The annual meeting of the shareholders (the "Annual Meeting") shall be held on the first Tuesday in February of each year at such time or on such other date as may be fixed by or under the authority of the Board of Directors. If the day fixed for the Annual Meeting shall be a legal holiday in the State of Wisconsin, then such meeting shall be held on the next succeeding Business Day (as defined below). In fixing a meeting date for any Annual Meeting, the Board of Directors may consider such factors as it deems relevant within the good faith exercise of its business judgment. At each Annual Meeting, the shareholders shall elect individuals to the Board of Directors in accordance with these by-laws. At any such Annual Meeting, only other business properly brought before the Annual Meeting in accordance with Section 2.11 may be transacted.

2.02 Special Meeting.

- (a) A special meeting of the shareholders (a "Special Meeting") may be called only by (i) a majority of the Board of Directors, (ii) the Chairman of the Board or (iii) the Chief Executive Officer. The Chief Executive Officer shall call a Special Meeting upon the demand, in accordance with this Section 2.02, of the holders of record representing at least 10% of all the votes entitled to be cast on any issue proposed to be considered at the Special Meeting.
- (b) To enable the corporation to determine the shareholders entitled to demand a Special Meeting, the Board of Directors may fix a record date to determine the shareholders entitled to make such a demand (the "Demand Record Date"). The Demand Record Date shall not precede the date on which the Board of Directors adopts the resolution fixing the Demand Record Date and shall not be more than ten days after the date on which the resolution fixing the Demand Record Date is adopted by the Board of Directors. Any shareholder of record entitled to demand a Special Meeting who is seeking to have shareholders demand a Special Meeting shall, by sending written notice to the Secretary at the principal offices of the corporation, by hand or by certified or registered mail, return receipt requested, request the Board of Directors to fix a Demand Record Date. The Board of Directors shall promptly, but in all events within ten days after the date on which a valid request to fix a Demand Record Date is received, adopt a resolution fixing the Demand Record Date and shall make a public announcement of such Demand Record Date. If no Demand Record Date has been fixed by the Board of Directors within ten days after the date on which such request is received by the Secretary at the principal offices of the corporation, then the Demand Record Date shall be the 10th day after the first date on which a valid written request to set a Demand Record Date is received by the Secretary at the principal offices of the corporation. To be valid, such written request shall set forth the purpose or purposes for which the Special Meeting is to be held, shall be signed by one or more shareholders of record and by the beneficial owner or owners, if any, on whose behalf the request is made, shall bear the date of signature of each such shareholder and any such beneficial owner and shall set forth all information that would be required to be set forth in a notice described in Section 2.11(a)(ii) delivered by such shareholder or shareholders as if the

- (c) For a shareholder or shareholders to demand a Special Meeting, a written demand or demands for a Special Meeting by the holders of record as of the Demand Record Date of shares representing at least 10% of all the votes entitled to be cast on any issue proposed to be considered at the Special Meeting, calculated as if the Demand Record Date were the record date for the Special Meeting, must be delivered to the Secretary at the principal offices of the corporation. To be valid, each written demand by a shareholder for a Special Meeting shall set forth the specific purpose or purposes for which the Special Meeting is to be held (which purpose or purposes shall be limited to the purpose or purposes set forth in the written request to set a Demand Record Date received by the corporation pursuant to Section 2.02(b)), shall be signed by one or more persons who as of the Demand Record Date are shareholders of record and by the beneficial owners, if any, on whose behalf the demand is made, shall bear the date of signature of each such shareholder and any such beneficial owner, shall set forth the name and address of each such shareholder (as they appear in the corporation's books) and any such beneficial owner signing such demand and the class or series and number of shares of the corporation that are owned of record and/or beneficially by each such shareholder and any such beneficial owner, shall be sent to the Secretary at the principal offices of the corporation, by hand or by certified or registered mail, return receipt requested, and shall be received by the Secretary at the principal offices of the corporation within seventy days after the Demand Record Date.
- (d) The corporation shall not be required to call a Special Meeting upon shareholder demand unless, in addition to the documents required by Section 2.02(c), the Secretary receives a written agreement signed by each Soliciting Shareholder (as defined below) pursuant to which each Soliciting Shareholder, jointly and severally, agrees to pay the corporation's costs of holding the Special Meeting, including the costs of preparing and mailing proxy materials for the corporation's own solicitation, provided that if each of the resolutions introduced by any Soliciting Shareholder at such meeting is adopted, and each of the individuals nominated by or on behalf of any Soliciting Shareholder for election as a director at such meeting is elected, then the Soliciting Shareholders shall not be required to pay such costs. For purposes of these by-laws, the following terms shall have the respective meanings set forth below:
- (i) "Affiliate" of any Person (as defined herein) shall mean any Person controlling, controlled by or under common control with such first Person.
- (ii) "Business Day" shall mean any day other than a Saturday, a Sunday or a day on which banking institutions in the State of Wisconsin are authorized or obligated by law or executive order to close.
- (iii) "<u>Participant</u>" shall have the meaning assigned to such term in paragraphs (a)(iii), (iv), (v) and (vi) of Instruction 3 to Item 4 of Schedule 14A of the rules promulgated under the Securities Exchange Act of 1934, as amended (the "Exchange Act").
- (iv) "<u>Person</u>" shall mean any individual, firm, corporation, partnership, joint venture, association, trust, unincorporated organization or other entity.
- (v) "<u>Proxy</u>" shall have the meaning assigned to such term in Rule 14a-1 promulgated under the Exchange Act (and, in such Rule 14a-1, a consent or authorization shall be interpreted to include signature on a demand for purposes of construing all the definitions in this Section 2.02(d)).
 - (vi) "Solicitation" shall have the meaning assigned to such term in Rule 14a-1 promulgated under the Exchange Act.

- (vii) "Soliciting Shareholder" shall mean, with respect to any Special Meeting demanded by a shareholder or shareholders, each of the following Persons:
- (A) if the number of shareholders signing the demand or demands of meeting delivered to the Secretary at the principal offices of the corporation pursuant to Section 2.02(c) is ten or fewer, each Person signing any such demand; or
- (B) if the number of shareholders signing the demand or demands of meeting delivered to the corporation pursuant to Section 2.02(c) is more than ten, each Person who either (I) was a Participant in any Solicitation of such demand or demands or (II) at the time of the delivery to the Secretary at the principal offices of the corporation of the documents described in Section 2.02(c) had engaged or intends to engage in any Solicitation of Proxies for use at such Special Meeting (other than a Solicitation of Proxies on behalf of the corporation).
- A "Soliciting Shareholder" shall also mean each Affiliate of a Soliciting Shareholder described in clause (A) or (B) above who is a member of such Soliciting Shareholder's "group" for purposes of Rule 13d-5(b) under the Exchange Act, and any other Affiliate of such a Soliciting Shareholder, if a majority of the directors then in office determines, reasonably and in good faith, that such Affiliate should be required to sign the written notice described in Section 2.02(c) and/or the written agreement described in this Section 2.02(d) to prevent the purposes of this Section 2.02 from being evaded.
- (e) Except as provided in the following sentence, any Special Meeting shall be held at such date and hour as may be designated by whichever of the Board of Directors, the Chairman of the Board or the Chief Executive Officer shall have called such meeting. In the case of any Special Meeting called by the Chief Executive Officer upon the demand of shareholders (a "Demand Special Meeting"), such meeting shall be held at such date and hour as may be designated by the Board of Directors; provided, however, that the date of any Demand Special Meeting shall be not more than seventy days after the Meeting Record Date (as defined in Section 2.05(a)); and provided further that in the event that the directors then in office fail to designate a date and hour for a Demand Special Meeting within ten days after the date that valid written demands for such meeting by the holders of record as of the Demand Record Date of shares representing at least 10% of all the votes entitled to be cast on any issue proposed to be considered at the Special Meeting, calculated as if the Demand Record Date were the record date for the Special Meeting, are delivered to the corporation (the "Delivery Date"), then such meeting shall be held at 10:00 A.M., local time, on the 100th day after the Delivery Date or, if such 100th day is not a Business Day, on the first preceding Business Day. In fixing a meeting date for any Special Meeting, the Board of Directors or the Chairman of the Board may consider such factors as it, he or she deems relevant within the good faith exercise of its, his or her business judgment, including, without limitation, the nature of the action proposed to be taken, the facts and circumstances surrounding any demand for such meeting and any plan of the Board of Directors, the Chairman of the Board or the Chief Executive Officer to call an Annual Meeting or Special Meeting for the conduct of related business.
- (f) The corporation may engage regionally or nationally recognized independent inspectors of elections to act as an agent of the corporation for the purpose of promptly performing a ministerial review of the validity of any purported written demand or demands for a Special Meeting received by the Secretary. For the purpose of permitting the inspectors to perform such review, no purported demand shall be deemed to have been delivered to the corporation until the earlier of (i) five Business Days following receipt by the Secretary of such purported demand and (ii) such date as the independent inspectors certify to the corporation that the valid demands received by the Secretary represent at least 10% of all the votes entitled to be cast on any issue proposed to be considered at the Special Meeting, calculated as if the Demand Record Date were the record date for the Special Meeting. Nothing contained in this Section 2.02(f) shall in any way be construed to suggest or imply that the Board of Directors or any shareholder shall not be entitled to contest the validity of any demand, whether during or after such five Business Day period, or to take any other action (including, without limitation, the commencement, prosecution or defense of any litigation with respect thereto).
- 2.03 <u>Place of Meeting</u>. The Board of Directors or the Chairman of the Board may designate any place, either within or without the State of Wisconsin, as the place of meeting for any Annual Meeting or Special Meeting. If no designation is made, then the place of meeting shall be the principal office of the corporation. Any meeting may be postponed or adjourned pursuant to Section 2.07 to reconvene at any place designated by vote of the Board of Directors or by the Chairman of the Board.

2.04 Notice of Meeting.

- (a) Written notice stating the place, day and hour of an Annual Meeting or Special Meeting shall be delivered not less than ten days nor more than seventy days before the date of the meeting (unless a different date is required by the law or the articles of incorporation), by or at the direction of the Chairman of the Board or the Secretary, to each shareholder of record entitled to vote at such meeting and to such other persons as are required by the Wisconsin Business Corporation Law. In the event of any Demand Special Meeting, such notice of meeting shall be sent prior to the later of (x) two days after the Meeting Record Date for such Demand Special Meeting and (y) thirty days after the Delivery Date. For purposes of this Section 2.04, notice by "electronic transmission" (as defined in the Wisconsin Business Corporation Law) is written notice. Written notice pursuant to this Section 2.04 shall be deemed to be effective (i) when mailed, if mailed postpaid and addressed to the shareholder's address shown in the corporation's current record of shareholders or (ii) when electronically transmitted to the shareholder in a manner authorized by the shareholder.
- (b) In the case of any Special Meeting, (i) the notice of meeting shall describe any business that the Board of Directors shall have theretofore determined to bring before the meeting and (ii) in the case of a Demand Special Meeting, the notice of meeting (A) shall describe any business set forth in the statement of purpose of the demands received by the corporation in accordance with Section 2.02, (B) shall contain all of the information required in the notice received by the corporation in accordance with Section 2.11(b) and (C) shall describe any business that the Board of Directors shall have theretofore determined to bring before the Demand Special Meeting. Except as otherwise provided in these bylaws, in the articles of incorporation or in the Wisconsin Business Corporation Law, the notice of an Annual Meeting need not include a description of the purpose or purposes for which the meeting is called.
- (c) If any Annual Meeting or Special Meeting is adjourned to a different date, time or place, then the corporation shall not be required to give notice of the new date, time or place if the new date, time or place is announced at the meeting before adjournment; provided, however, that if a new Meeting Record Date for an adjourned meeting is or must be fixed, then the corporation shall give notice of the adjourned meeting to persons who are shareholders as of the new Meeting Record Date.

2.05 Fixing of Record Date.

- (a) The Board of Directors may fix in advance a date not less than ten days and not more than seventy days prior to the date of an Annual Meeting or Special Meeting as the record date for the determination of shareholders entitled to notice of, or to vote at, such meeting (the "Meeting Record Date"). In the case of any Demand Special Meeting, (i) the Meeting Record Date shall be not later than the 30th day after the Delivery Date and (ii) if the Board of Directors fails to fix the Meeting Record Date within thirty days after the Delivery Date, then the close of business on such 30th day shall be the Meeting Record Date. The shareholders of record on the Meeting Record Date shall be the shareholders entitled to notice of and to vote at the Annual Meeting or Special Meeting. When a determination of shareholders entitled to notice of or to vote at the Annual Meeting or Special Meeting has been made as provided in this section, such determination shall be applied to any adjournment thereof unless the Board of Directors fixes a new Meeting Record Date and except as otherwise required by law. A new Meeting Record Date must be set if a meeting is adjourned to a date more than one hundred twenty days after the date fixed for the original meeting.
- (b) The Board of Directors may also fix in advance a date as the record date for the purpose of determining shareholders entitled to take any other action or determining shareholders for any other purpose other than those set forth in Section 2.02(a) and Section 2.05(a). Such record date shall not be more than seventy days prior to the date on which the particular action, requiring such determination of shareholders, is to be taken. If the Board of Directors does not fix a record date for the determination of shareholders entitled to receive a share dividend or distribution (other than a distribution involving a purchase, redemption or other acquisition of the corporation's shares), then the close of business on the date on which the resolution of the Board of Directors is adopted declaring the dividend or distribution shall be the record date.

2.06 <u>Voting Records</u>. After a Meeting Record Date has been fixed, the corporation shall prepare a list of the names of all of the shareholders entitled to notice of the meeting. The shareholders' list shall be arranged by class or series of shares, if any, and show the address of and number of shares held by each shareholder. Any shareholder or his, her or its agent or attorney, on written demand, may inspect the shareholders' list beginning two Business Days after the corporation gives the notice of the meeting for which the shareholders' list was prepared and continuing to the date of the meeting, at the corporation's principal office or at a place identified in the meeting notice in the city where the meeting will be held, and if and to the extent entitled to do so under the Wisconsin Business Corporation Law, may copy the shareholders' list, during regular business hours and at his, her or its expense, during the period that it is available for inspection hereunder. The corporation shall make the shareholders' list available at the meeting and any shareholder or his, her or its agent or attorney may inspect the shareholders' list at any time during the meeting or any adjournment thereof. The original stock transfer books of the corporation shall be prima facie evidence as to who are the shareholders entitled to inspect the shareholders' list or to vote at any meeting of the shareholders. Refusal or failure to prepare or make available the shareholders' list shall not affect the validity of any action taken at a meeting of shareholders.

2.07 Quorum; Postponement; Adjournments.

- (a) Shares entitled to vote as a separate voting group may take action on a matter at an Annual Meeting or Special Meeting only if a quorum of those shares exists with respect to that matter. Except as otherwise provided in the articles of incorporation or in the Wisconsin Business Corporation Law, a majority of the votes entitled to be cast on a matter by the voting group shall constitute a quorum of that voting group for action on that matter. Once a share is represented for any purpose at an Annual Meeting or Special Meeting, other than for the purpose of objecting to holding the meeting or transacting business at the meeting, it is considered present for purposes of determining whether a quorum exists for the remainder of the meeting and for any adjournment of the meeting unless a new Meeting Record Date is or must be set for the adjourned meeting. If a quorum exists, then action on a matter, other than the election of directors, by a voting group is approved if the votes cast within the voting group favoring the action exceed the votes cast opposing the action, unless the articles of incorporation or the Wisconsin Business Corporation Law requires a greater number of affirmative votes. Unless otherwise provided in the articles of incorporation, each director shall be elected by a plurality of the votes cast by the shares entitled to vote in the election of directors at a meeting at which a quorum is present.
- (b) The Board of Directors acting by resolution may postpone and reschedule any previously scheduled Annual Meeting or Special Meeting; provided, however, that a Demand Special Meeting shall not be postponed beyond the 100th day following the Delivery Date. Any Annual Meeting or Special Meeting may be adjourned from time to time, whether or not there is a quorum, (i) at any time, upon a resolution by shareholders if the votes cast in favor of such resolution by the holders of shares of each voting group entitled to vote on any matter theretofore properly brought before the meeting exceed the number of votes cast against such resolution by the holders of shares of each such voting group or (ii) at any time prior to the transaction of any business at such meeting, by the Chairman of the Board or pursuant to a resolution of the Board of Directors; provided, however, that a Demand Special Meeting adjourned pursuant to clause (ii) must be reconvened on or before the 100th day following the Delivery Date. No notice of the time and place of adjourned meetings need be given except as required by the Wisconsin Business Corporation Law. At any adjourned meeting at which a quorum shall be present or represented, any business may be transacted that might have been transacted at the meeting as originally noticed.
- Conduct of Meetings. The Chairman of the Board or, in his or her absence, the Chief Executive Officer or, in the Chief Executive Officer's absence, the President (if the Chief Executive Officer is not then the President) or a Vice President designated by the Board of Directors shall call any Annual Meeting or Special Meeting to order and shall act as chairman of the meeting, and the Secretary shall act as secretary of all meetings of the shareholders, but, in the absence of the Secretary, the presiding officer may appoint any other person to act as secretary of the meeting. The Board of Directors may, to the extent not prohibited by law, adopt by resolution such rules and regulations for the conduct of an Annual Meeting or Special Meeting as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board of Directors, the chairman of the meeting shall have the right and authority to prescribe such rules, regulations or procedures and to do such acts as, in the judgment of the chairman of the meeting, are appropriate for the proper conduct of an Annual Meeting or Special Meeting. Such rules, regulations or procedures, whether adopted by the Board of Directors or prescribed by the chairman of the meeting, may to the extent not prohibited by law include, without limitation, the following: (a) the establishment of an agenda or order of business for the meeting; (b) rules and procedures for maintaining order at the meeting and the safety of those present; (c) limitations on attendance at or participation in the meeting to shareholders of record of the corporation, their duly authorized and constituted proxies (which shall be reasonable in number) or such other persons as the chairman of the meeting shall determine; (d) restrictions on entry to the meeting after the time fixed for the commencement thereof; (e) limitations on the time allotted to questions or comments by participants; (f) rules and procedures regarding the execution of election ballots before or after the time fixed for the commencement of the meeting; (g) the appointment of an inspector of election or an officer or agent of the corporation authorized to tabulate votes; and (h) rules and procedures to facilitate the conduct of, and participation in, the meeting by electronic means.

- 2.09 <u>Proxies</u>. At any Annual Meeting or Special Meeting, a shareholder entitled to vote may vote in person or by proxy. A shareholder entitled to vote at any Annual Meeting or Special Meeting may authorize another person to act for the shareholder by appointing the person as a proxy. The means by which a shareholder or the shareholder's authorized officer, director, employee, agent or attorney-in-fact may authorize another person to act for the shareholder by appointing the person as proxy include:
- (a) Appointment of a proxy in writing by signing or causing the shareholder's signature to be affixed to an appointment form by any reasonable means, including, without limitation, by facsimile signature.
- (b) Appointment of a proxy by transmitting or authorizing the transmission of an electronic transmission of the appointment to the person who will be appointed as proxy or to a proxy solicitation firm, proxy support service organization or like agent authorized to receive the transmission by the person who will be appointed as proxy. Every electronic transmission shall contain, or be accompanied by, information that can be used to reasonably determine that the shareholder transmitted or authorized the transmission of the electronic transmission. Any person charged with determining whether a shareholder transmitted or authorized the transmission of the electronic transmission shall specify the information upon which the determination is made.

An appointment of a proxy is effective when a signed appointment form or an electronic transmission of the appointment is received by the inspector of election or the officer or agent of the corporation authorized to tabulate votes. Unless the appointment form or electronic transmission states that the proxy is irrevocable and the appointment is coupled with an interest, a proxy may be revoked at any time before it is voted, either by written notice filed with the Secretary or the secretary of the meeting or by oral notice given by the shareholder to the presiding officer during the meeting. The presence of a shareholder who has made an effective proxy appointment shall not of itself constitute a revocation. A proxy appointment is valid for eleven months unless a different period is expressly provided in the appointment. The Board of Directors, the Chairman of the Board and the Chief Executive Officer each shall have the power and authority to make rules as to the validity and sufficiency of proxies.

- 2.10 <u>Voting of Shares</u>. Each outstanding share shall be entitled to one vote on each matter submitted to a vote at any Annual Meeting or Special Meeting, except to the extent that the voting rights of the shares of any class or classes are enlarged, limited or denied by the articles of incorporation.
 - 2.11 Notice of Shareholder Business and Nomination of Directors.
 - (a) Annual Meetings.
- (i) Nominations of persons for election to the Board of Directors and the proposal of other business to be considered by the shareholders may be made at an Annual Meeting (A) pursuant to the corporation's notice of meeting, (B) by or at the direction of the Board of Directors, (C) by any shareholder of the corporation who (1) is a shareholder of record at the time of giving of notice provided for in this Section 2.11, (2) is entitled to vote with respect to such nomination or other business at the meeting under the articles of incorporation and (3) complies with the notice procedures set forth in this Section 2.11 or (D) with respect to nominations, by any shareholder of the corporation who is eligible under, and complies with the notice procedures set forth in, Section 2.15. Clause (C) and clause (D) in the preceding sentence shall be the exclusive means for a shareholder to make nominations or submit other business (other than matters properly brought under Rule 14a-8 under the Exchange Act and included in the corporation's notice of meeting) before an Annual Meeting.

For nominations or other business to be properly brought before an Annual Meeting by a shareholder pursuant to Section 2.11(a)(i)(C), the shareholder must have given timely notice thereof in writing to the Secretary and such other business must otherwise be a proper matter for shareholder action. To be timely, a shareholder's notice shall be received by the Secretary at the principal offices of the corporation not less than forty-five days nor more than seventy days prior to the first annual anniversary of the date set forth in the corporation's proxy statement for the immediately preceding Annual Meeting as the date on which the corporation first made available to its shareholders definitive proxy materials for the immediately preceding Annual Meeting (the "Anniversary Date"); provided, however, that if the date for which the Annual Meeting is called is more than thirty days before or more than thirty days after the first annual anniversary of the immediately preceding Annual Meeting, then notice by the shareholder to be timely must be received by the Secretary not earlier than the close of business on the 100th day prior to the date of such Annual Meeting and not later than the later of (A) the 75th day prior to the date of such Annual Meeting or (B) the 10th day following the day on which public announcement of the date of such Annual Meeting is first made. In no event shall any adjournment or postponement of an Annual Meeting or the announcement thereof commence a new time period for the giving of a shareholder notice as described above. To be in proper form, a shareholder's notice to the Secretary (whether given pursuant to this Section 2.11(a)(ii) or Section 2.02(b)) shall be signed by the shareholder of record who intends to make the nomination or introduce the other business and by the beneficial owner or owners, if any, on whose behalf the shareholder is acting, shall bear the date of signature of such shareholder and any such beneficial owner and shall set forth: (I) the name and address of such shareholder (as they appear on the corporation's books) and any such beneficial owner; (II) the Share Information relating to each such shareholder and beneficial owner (which Share Information shall be supplemented by such shareholder and any such beneficial owner not later than ten days after the Meeting Record Date to disclose such Share Information as of the Meeting Record Date); (III) a representation that such shareholder is a holder of record of shares of the corporation entitled to vote under the articles of incorporation at such meeting with respect to such nomination or other business and intends to appear in person or by proxy at the meeting to make such nomination or introduce such other business; (IV) any other information relating to such shareholder and any such beneficial owner that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal and/or for the election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder; (V) in the case of any proposed nomination for election or re-election as a director, (1) the name and residence address of the person or persons to be nominated, (2) a description of all agreements, arrangements or understandings between such shareholder and any such beneficial owner and each nominee and any other person or persons (naming such person or persons) pursuant to which the nomination is to be made by such shareholder and any such beneficial owner, including without limitation any agreement, arrangement or understanding with any person as to how such nominee, if elected as a director of the corporation, will act or vote on any issue or question, (3) a description of all direct and indirect compensation and other material agreements, arrangements and understandings during the past three years and any other material relationships, between or among such shareholder and any such beneficial owner and their respective Affiliates and associates, or others acting in concert therewith, on the one hand, and each proposed nominee, and his or her respective Affiliates and associates, or others acting in concert therewith, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 of Regulation S-K of the Securities and Exchange Commission (the "SEC") if the shareholder making the nomination and any beneficial owner on whose behalf the nomination is made, or any Affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of such rule and the nominee were a director or executive officer of such registrant, (4) such other information regarding each nominee proposed by such shareholder and any such beneficial owner as would be required to be disclosed by such shareholder and any such beneficial owner in contested solicitations of proxies for elections of directors, or would be otherwise required to be disclosed, in each case pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder, and (5) the written consent of each nominee proposed by such shareholder and any such beneficial owner to be named in a proxy statement and to serve as a director of the corporation if so elected: (VI) in the case of any proposed removal of a director at a meeting called for the purpose of removing such director. (1) the names of the directors to be removed and (2) the reasons of such shareholder and any such beneficial owner for asserting that such directors should be removed; and (VII) in the case of any other business that such shareholder and any such beneficial owner propose to bring before the meeting, (1) a brief description of the business desired to be brought before the meeting and, if such business includes a proposal to amend these by-laws, the language of the proposed amendment, (2) the reasons of such shareholder and any such beneficial owner for conducting such business at the meeting, (3) any material interest in such business of such shareholder and any such beneficial owner and (4) a description of all agreements, arrangements or understandings between such shareholder and any such beneficial owner and any other person or persons (naming such person or persons) in connection with the proposal of such business by such shareholder. In the case of any proposed nomination for election or reelection as a director, the corporation may require any proposed nominee to furnish such other information as may reasonably be required by the corporation to determine the eligibility of such proposed nominee to serve as an independent director of the corporation or that could be material to a reasonable shareholder's understanding of the independence, or lack thereof, of such nominee.

For purposes of these by-laws, the term "Share Information" shall mean (1) the class or series and number of shares of the corporation that are owned, directly or indirectly, of record and/or beneficially by a shareholder, any beneficial owner on whose behalf the shareholder is acting and any of their respective Affiliates, (2) any option, warrant, convertible security, stock appreciation right or similar right with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class or series of shares of the corporation or with a value derived in whole or in part from the value of any class or series of shares of the corporation, whether or not such instrument or right shall be subject to settlement in the underlying class or series of capital stock of the corporation or otherwise (a "Derivative Instrument") directly or indirectly owned beneficially by such shareholder, any such beneficial owner and any of their respective Affiliates, and any other direct or indirect opportunity to profit or share in any profit derived from any increase or decrease in the value of shares of the corporation, (3) any proxy, agreement, arrangement, understanding, or relationship pursuant to which such shareholder has a right to vote any shares of any security of the corporation, (4) any short interest in any security of the corporation (for purposes of these by-laws, a person shall be deemed to have a short interest in a security if such person directly or indirectly, through any agreement, arrangement, understanding, relationship or otherwise, has the opportunity to profit or share in any profit derived from any decrease in the value of the subject security), (5) any rights to dividends on the shares of the corporation owned beneficially by such shareholder that are separated or separable from the underlying shares of the corporation, (6) any proportionate interest in shares of the corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership in which such shareholder or beneficial owner is a general partner or, directly or indirectly, beneficially owns an interest in a general partner and (7) any performance-related fees (other than asset-based fees) to which such shareholder, any such beneficial owner or any of their respective Affiliates are entitled based on any increase or decrease in the value of shares of the corporation or Derivative Instruments, if any, as of the date of such notice, including without limitation any such interests held by members of such person's immediate family as defined in Item 404 of Regulation S-

(iii) Notwithstanding anything in the second sentence of Section 2.11(a)(ii) to the contrary, if the number of directors to be elected to the Board of Directors is increased and there is no public announcement naming all of the nominees for director or specifying the size of the increased Board of Directors made by the corporation at least forty-five days prior to the Anniversary Date, then a shareholder's notice required by this Section 2.11 shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be received by the Secretary at the principal offices of the corporation not later than the close of business on the 10th day following the day on which such public announcement is first made by the corporation.

Special Meetings. Only such business shall be conducted at a Special Meeting as shall have been described in the corporation's notice of meeting sent to shareholders pursuant to Section 2.04. Nominations of persons for election to the Board of Directors may be made at a Special Meeting at which directors are to be elected pursuant to the corporation's notice of meeting (i) by or at the direction of the Board of Directors or (ii) by any shareholder of the corporation who (A) is a shareholder of record at the time of giving of such notice of meeting, (B) is entitled to vote with respect to such nominations at the meeting under the articles of incorporation and (C) complies with the notice procedures set forth in this Section 2.11. Clause (ii) in the preceding sentence shall be the exclusive means for a shareholder to make nominations before a Special Meeting. Any shareholder permitted to nominate persons for election to the Board of Directors pursuant to such clause (ii) who desires to nominate persons for election to the Board of Directors at such a Special Meeting shall cause a written notice to be received by the Secretary at the principal offices of the corporation not earlier than ninety days prior to such Special Meeting and not later than the close of business on the later of (I) the 60th day prior to such Special Meeting and (II) the 10th day following the day on which public announcement is first made of the date of such Special Meeting and of the nominees proposed by the Board of Directors to be elected at such meeting. Such written notice shall be signed by the shareholder of record who intends to make the nomination and by the beneficial owner or owners, if any, on whose behalf the shareholder is acting, shall bear the date of signature of such shareholder and any such beneficial owner and shall set forth: (1) the name and address of such shareholder (as they appear in the corporation's books) and any such beneficial owner; (2) the Share Information relating to each such shareholder and beneficial owner (which Share Information shall be supplemented by such shareholder and any such beneficial owner not later than ten days after the Meeting Record Date to disclose such Share Information as of the Meeting Record Date); (3) a representation that such shareholder is a holder of record of shares of the corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to make the nomination specified in the notice; (4) any other information relating to such shareholder and any such beneficial owner that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for the election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder; (5) the name and residence address of the person or persons to be nominated; (6) a description of all agreements, arrangements or understandings between such shareholder and any such beneficial owner and each nominee and any other person or persons (naming such person or persons) pursuant to which the nomination is to be made by such shareholder and any such beneficial owner, including without limitation any agreement, arrangement or understanding with any person as to how such nominee, if elected as a director of the corporation, will act or vote on any issue or question; (7) a description of all direct and indirect compensation and other material agreements, arrangements and understandings during the past three years and any other material relationships, between or among such shareholder and any such beneficial owner and their respective Affiliates and associates, or others acting in concert therewith, on the one hand, and each proposed nominee, and his or her respective Affiliates and associates, or others acting in concert therewith, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 of Regulation S-K of the SEC if the shareholder making the nomination and any beneficial owner on whose behalf the nomination is made, or any Affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of such rule and the nominee were a director or executive officer of such registrant; and (8) the written consent of each nominee proposed by such shareholder and any such beneficial owner to be named in a proxy statement and to serve as a director of the corporation if so elected. In no event shall any adjournment or postponement of a Special Meeting commence a new time period for the giving of a shareholder's notice as described above.

(c) General.

- (i) Only persons who are nominated in accordance with the procedures set forth in this Section 2.11 or Section 2.15 shall be eligible to be elected as directors at an Annual Meeting or Special Meeting. Only such business shall be conducted at an Annual Meeting or Special Meeting as shall have been brought before such meeting in accordance with the procedures set forth in this Section 2.11. The chairman of the meeting shall have the power and duty to determine whether a nomination or any business proposed to be brought before the meeting was made in accordance with the procedures set forth in this Section 2.11 or Section 2.15 and, if any proposed nomination or business is not in compliance with this Section 2.11 or Section 2.15, as the case may be, to declare that such defective nomination or proposal shall be disregarded.
- (ii) For purposes of this Section 2.11, "public announcement" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press or comparable national news service or in a document publicly filed by the corporation with the SEC pursuant to Section 13, 14 or 15(d) of the Exchange Act.
- (iii) Notwithstanding the foregoing provisions of this Section 2.11, a shareholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 2.11; provided, however, that any references in these by-laws to the Exchange Act or the rules and regulations promulgated thereunder are not intended to limit the requirements applicable to nominations or shareholder action pursuant to Sections 2.11(a)(ii) or 2.11(b). Nothing in this Section 2.11 shall be deemed to limit the corporation's obligation to include shareholder proposals in its proxy statement if such inclusion is required by Rule 14a-8 under the Exchange Act.

2.12 <u>Voting of Shares by Certain Holders.</u>

(a) Other Corporations. Shares standing in the name of another corporation may be voted, either in person or by proxy, by the president of such other corporation or any other officers appointed by such president. An appointment of a proxy executed by any principal officer of such other corporation or assistant thereto shall be conclusive evidence of the signer's authority to act, in the absence of express notice to the contrary, given in writing by the Board of Directors of such other company to the Secretary.

- (b) <u>Legal Representatives and Fiduciaries</u>. Shares held by an administrator, executor, guardian, conservator, trustee in bankruptcy, receiver or assignee for creditors may be voted by such person, either in person or by proxy, without a transfer of such shares into such person's name, provided that there is filed with the Secretary before or at the time of the meeting proper evidence of such person's incumbency and the number of shares held. Shares held by a fiduciary may be voted by the person acting in such capacity, either in person or by proxy. A proxy executed by a fiduciary, shall be conclusive evidence of the signer's authority to act, in the absence of express notice to the contrary, given in writing to the Secretary.
- (c) <u>Pledges</u>. A shareholder whose shares are pledged shall be entitled to vote such shares until the shares have been transferred into the name of the pledgee, and thereafter the pledgee shall be entitled to vote the shares so transferred.
- (d) <u>Treasury Stock and Subsidiaries</u>. Neither treasury shares nor shares held by another corporation if a majority of the shares entitled to vote for the election of directors of such other corporation is held by this corporation shall be voted at any Annual Meeting or Special Meeting or counted in determining the total number of outstanding shares entitled to vote, but shares of its own issue held by this corporation in a fiduciary capacity, or held by such other corporation in a fiduciary capacity, shall be entitled to vote and shall be counted in determining the total number of outstanding shares entitled to vote.
- (e) Minors. Shares held by a minor may be voted by such minor in person or by proxy, and no such vote shall be subject to disaffirmance or avoidance, unless, prior to such vote, the Secretary has received written notice or has actual knowledge that such shareholder is a minor.
- (f) <u>Incompetents and Spendthrifts</u>. Shares held by a incompetent or spendthrift may be voted by such incompetent or spendthrift in person or by proxy, and no such vote shall be subject to disaffirmance or avoidance, unless, prior to such vote, the Secretary has actual knowledge that such shareholder has been adjudicated an incompetent or spendthrift or has actual knowledge of filing of judicial proceedings for appointment of a guardian.
- (g) <u>Joint Tenants</u>. Shares registered in the names of two or more individuals who are named in the registration as joint tenants may be voted in person or by proxy signed by any one or more of such individuals if either (i) no other such individual or his or her legal representative is present at the meeting and claims the right to participate in the voting of such shares or prior to the vote filed with the Secretary a contrary written voting authorization or direction or written denial of authority of the individual present or signing the proxy proposed to be voted or (ii) all such other individuals are deceased and the Secretary has no actual knowledge that the survivor has been adjudicated not to be the successor to the interest of those deceased.
- 2.13 Waiver of Notice by Shareholders. Whenever any notice is required to be given to any shareholder of the corporation under the articles of incorporation, these by-laws or any provision of law, a waiver thereof in writing, signed at any time, whether before or after the time of the meeting, and delivered to the Secretary for inclusion in the corporation's records, by the shareholder entitled to such notice, shall be deemed equivalent to the giving of such notice; provided that such waiver in respect to any matter of which notice is required under any provision of the Wisconsin Business Corporation Law shall contain the same information as would have been required to be included in such notice, except the date, time and place of the meeting. A shareholder's attendance at any Annual Meeting or Special Meeting, in person or by proxy, waives objection to all of the following: (a) lack of notice or defective notice of the meeting, unless the shareholder, at the beginning of the meeting or promptly upon arrival, objects to holding the meeting or transacting business at the meeting; and (b) consideration of a particular matter at the meeting that is not within the purpose described in the meeting notice, unless the shareholder objects to considering the matter when it is presented.
- 2.14 <u>Unanimous Consent Without Meeting</u>. Any action required or permitted by the articles of incorporation, these by-laws or any provision of law to be taken at an Annual Meeting or Special Meeting may be taken without a meeting if a consent in writing, setting forth the action so taken, shall be signed by all of the shareholders entitled to vote with respect to the subject matter thereof and delivered to the Secretary for inclusion in the corporation's records.

2.15	Shareholder	Nominations	Included in the	Corporation'	's Proxy	y Materials
2.13	Shareholder	Nommanons	menuded in the	Corporation	S FIUX	y iviatella

- (a) <u>Inclusion of Nominee in Proxy Statement</u>. Subject to the provisions of this Section 2.15, if expressly requested in the relevant Nomination Notice (as defined below), the corporation shall include in its proxy statement for any Annual Meeting:
- (i) the name of any person nominated for election (the "Nominee") to the Board of Directors, which shall also be included on the corporation's form of proxy and ballot for the relevant Annual Meeting, by any Eligible Holder (as defined below) or group of up to 20 Eligible Holders that has (individually and collectively, in the case of a group) satisfied, as determined by the Board of Directors or its designee, acting in good faith, all applicable conditions and complied with all applicable procedures set forth in this Section 2.15 (such Eligible Holder or group of Eligible Holders being a "Nominating Shareholder");
- (ii) disclosure about the Nominee and the Nominating Shareholder required under SEC rules or any other applicable law, rule or regulation to be included in the proxy statement; and
- (iii) any statement included by the Nominating Shareholder in the Nomination Notice for inclusion in the proxy statement in support of the Nominee's election to the Board of Directors (subject, without limitation, to Section 2.15(e)(ii)), if such statement does not exceed 500 words.

Notwithstanding anything herein to the contrary, the corporation may solicit shareholders against any Nominee and include in its proxy statement for any Annual Meeting any other information that the corporation or the Board of Directors determines, in their discretion, to include in the proxy statement relating to the nomination of the Nominee, including without limitation any statement in opposition to the nomination and any of the information provided pursuant to this Section 2.15.

(b) Maximum Number of Nominees.

- (i) The corporation shall not be required to include in the proxy statement for an Annual Meeting more Nominees than that number of directors constituting 20% of the total number of directors of the corporation on the last day on which a Nomination Notice may be submitted pursuant to Section 2.15(d) (the "Final Nomination Date"), rounded down to the nearest whole number, but not less than two (the "Maximum Number"). The Maximum Number for a particular Annual Meeting shall be reduced by (A) Nominees nominated by a Nominating Shareholder for that Annual Meeting whose nomination is subsequently withdrawn after the Nominating Shareholder is notified by the corporation that the Nominees will be included in the corporation's proxy statement and proxy card for the Annual Meeting, (B) Nominees nominated by a Nominating Shareholder for such Annual Meeting pursuant to this Section 2.15 that the Board of Directors itself decides to nominate for election at such Annual Meeting and (C) the number of directors in office as of the Final Nomination Date who had been Nominees nominated by a Nominating Shareholder with respect to any of the preceding two Annual Meetings (including any Nominee who had been counted at any such Annual Meeting pursuant to the immediately preceding clause (B)) and whose reelection at the upcoming Annual Meeting is being recommended by the Board of Directors. If one or more vacancies for any reason occurs on the Board of Directors after the Final Nomination Date but before the date of the Annual Meeting and the Board of Directors resolves to reduce the size of the Board of Directors in office as so reduced.
- (ii) Any Nominating Holder submitting more than one Nominee pursuant to this Section 2.15 for an Annual Meeting shall rank such Nominees based on the order in which the Nominating Holder desires such Nominees to be selected for inclusion in the corporation's proxy statement for such Annual Meeting if the number of Nominees pursuant to this Section 2.15 exceeds the Maximum Number. If the number of Nominees pursuant to this Section 2.15 for any Annual Meeting exceeds the Maximum Number, then the highest ranking Nominee who meets the requirements of this Section 2.15 from each Nominating Holder will be selected for inclusion in the corporation's proxy statement until the Maximum Number is reached, going in order of the amount (largest to smallest) of the shares of common stock of the corporation disclosed as owned in each Nominating Shareholder's Nomination Notice.

If, after the Final Nomination Date, (A) the corporation is notified, or the Board of Directors or its designee, acting in good faith, determines that a Nominating Shareholder has failed to satisfy or to continue to satisfy the eligibility requirements described in Section 2.15(c), any of the representations and warranties made in the Nomination Notice cease to be true and accurate in all material respects (or omit a material fact necessary to make the statements therein not misleading) or any material violation or breach occurs of the obligations, agreements, representations or warranties of the Nominating Shareholder or the Nominee under this Section 2.15, (B) a Nominating Shareholder or any qualified representative thereof does not appear at the Annual Meeting to present any nomination submitted pursuant to this Section 2.15, or the Nominating Shareholder withdraws its nomination, or (C) a Nominee becomes ineligible for inclusion in the corporation's proxy statement pursuant to this Section 2.15 or dies, becomes disabled or is otherwise disqualified from being nominated for election or serving as a director of the corporation or is unwilling or unable to serve as a director of the corporation, in each case as determined by the Board of Directors or its designee, acting in good faith, whether before or after the corporation's definitive proxy statement for such Annual Meeting is made available to shareholders, then the nomination of the Nominating Shareholder or such Nominee, as the case may be, shall be disregarded and no vote on such Nominee will occur (notwithstanding that proxies in respect of such vote may have been received by the corporation), the Nominating Shareholder may not cure in any way any defect preventing the nomination of the Nominee, and the corporation (1) may omit from its proxy statement and any ballot or form of proxy the disregarded Nominee and any information concerning such Nominee (including a Nominating Shareholder's statement in support) or any successor or replacement nominee proposed by the Nominating Shareholder or by any other Nominating Shareholder and (2) may otherwise communicate to its shareholders, including without limitation by amending or supplementing its proxy statement or ballot or form of proxy, that the Nominee will not be included as a Nominee in the proxy statement or on any ballot or form of proxy and will not be voted on at the Annual Meeting.

(c) <u>Eligibility of Nominating Shareholder.</u>

- (i) An "Eligible Holder" is a person who has either (A) been a record holder of the shares of the corporation's common stock used to satisfy the eligibility requirements in this Section 2.15(c) continuously for the three-year period specified in Section 2.15(c) (ii) or (B) provides to the Secretary of the corporation, within the time period referred to in Section 2.15(d), evidence of continuous ownership of such shares for such three-year period from one or more securities intermediaries in a form and in substance that the Board of Directors or its designee, acting in good faith, determines would be deemed acceptable for purposes of a shareholder proposal under Rule 14a-8(b)(2) under the Exchange Act (or any successor rule).
- (ii) An Eligible Holder or group of up to 20 Eligible Holders may submit a nomination in accordance with this Section 2.15 only if the person or group (in the aggregate) has continuously owned at least the Minimum Number (as defined below) of shares of the corporation's common stock throughout the three-year period preceding and including the date of submission of the Nomination Notice and continues to own at least the Minimum Number through the date of the Annual Meeting. A group of funds under common management and investment control shall be treated as one Eligible Holder for purposes of such limitation if such Eligible Holder shall provide together with the Nomination Notice documentation reasonably satisfactory to the corporation that demonstrates that the funds are under common management and investment control. For the avoidance of doubt, in the event of a nomination by a group of Eligible Holders, any and all requirements and obligations applicable to an individual Eligible Holder that are set forth in this Section 2.15, including the minimum holding period, shall apply to each member of such group; provided, however, that the Minimum Number shall apply to the ownership of the group in the aggregate, and a breach of any obligation, agreement, representation or warranty under this Section 2.15 by any member of a group shall be deemed a breach by the Nominating Shareholder. If any shareholder withdraws from a group of Eligible Holders at any time prior to the Annual Meeting, then the group of Eligible Shareholders shall only be deemed to own the shares held by the remaining members of the group and if, as a result of such withdrawal, the Nominating Shareholder no longer owns the Minimum Number of shares of the corporation's common stock, then the nomination shall be disregarded as provided in Section 2.15(b)(iii).

- (iii) The "Minimum Number" of shares of the corporation's common stock means 3% of the number of outstanding shares of the corporation's common stock as of the most recent date for which such amount is given in any filing by the corporation with the SEC prior to the submission of the Nomination Notice.
- (iv) For purposes of this Section 2.15, an Eligible Holder "owns" only those outstanding shares of common stock of the corporation as to which the Eligible Holder possesses both:
 - (A) the full voting and investment rights pertaining to such shares; and
 - (B) the full economic interest in (including the opportunity for profit and risk of loss on) such shares;

provided that the number of shares calculated in accordance with clauses (A) and (B) shall not include any shares (1) sold by such Eligible Holder or any of its affiliates in any transaction that has not been settled or closed, (2) borrowed by such Eligible Holder or any of its affiliates for any purpose or purchased by such Eligible Holder or any of its affiliates pursuant to an agreement to resell or (3) subject to any option, warrant, forward contract, swap, contract of sale, other derivative or similar instrument or agreement entered into by such Eligible Holder or any of its affiliates, whether any such instrument or agreement is to be settled with shares or with cash based on the notional amount or value of outstanding shares of common stock of the corporation, in any such case which instrument or agreement has, or is intended to have, the purpose or effect of (x) reducing in any manner, to any extent or at any time in the future, such Eligible Holder's or any of its affiliates' full right to vote or direct the voting of any such shares and/or (y) hedging, offsetting, or altering to any degree gain or loss arising from maintaining the full economic ownership of such shares by such Eligible Holder or any of its affiliates. An Eligible Holder "owns" shares held in the name of a nominee or other intermediary so long as the Eligible Holder retains the right to instruct how the shares are voted with respect to the election of directors and possesses the full economic interest in the shares. An Eligible Holder's ownership of shares shall be deemed to continue during any period in which the Eligible Holder has delegated any voting power by means of a proxy, power of attorney or other similar instrument or arrangement that is revocable at any time by the Eligible Holder. An Eligible Holder's ownership of shares shall be deemed to continue during any period in which the Eligible Holder has loaned such shares provided that the Eligible Holder has the power to recall such loaned shares on five business days' notice, recalls such loaned shares upon being notified by the corporation that any of the Eligible Holder's Nominees will be included in the corporation's proxy statement and proxy card for the Annual Meeting (subject to the provisions of this Section 2.15) and holds such shares through the date of the Annual Meeting. The terms "owned," "owning," "ownership" and other variations of the word "own" shall have correlative meanings. Whether outstanding shares of the corporation are "owned" for these purposes shall be determined by the Board. For purposes of this Section 2.15, the term "affiliate" or "affiliates" shall have the meaning ascribed thereto under the General Rules and Regulations under the Exchange Act.

- (v) No person shall be permitted to be in more than one group constituting a Nominating Shareholder, and if any person appears as a member of more than one group, then it shall be deemed to be a member of the group that has the largest amount of shares of common stock of the corporation disclosed as owned in the Nomination Notice.
- (d) Nomination Notice. To nominate a Nominee for purposes of this Section 2.15, the Nominating Shareholder must have given timely notice thereof in writing to the Secretary. To be timely, a Nominating Shareholder's notice shall be received by the Secretary at the principal offices of the corporation not less than 120 days nor more than 150 days prior to the first annual anniversary of the date set forth in the corporation's proxy statement for the immediately preceding Annual Meeting as the date on which the corporation first made available to its shareholders definitive proxy materials for the immediately preceding Annual Meeting; provided, however, that if the date for which the Annual Meeting is called is more than thirty days before or more than thirty days after the first annual anniversary of the immediately preceding Annual Meeting, then notice by the Nominating Shareholder to be timely must be received by the Secretary by the later of the close of business on the date that is 180 days prior to the date of such Annual Meeting or the tenth day following the day on which public announcement of such Annual Meeting is first made. In no event shall any adjournment or postponement of any Annual Meeting or the announcement thereof commence a new time period for the giving of a Nomination Notice. To be in proper form, a Nominating Shareholder's notice to the Secretary for purposes of this Section 2.15 shall include all of the following information and documents (collectively, the "Nomination Notice"):

(i) A Schedule 14N (or any successor form) relating to the Nominee, completed and filed with the SEC by the Nominating Shareholder as applicable, in accordance with SEC rules;
(ii) A written notice of the nomination of such Nominee that includes the following additional information, agreements, representations and warranties by the Nominating Shareholder (including each group member):
(A) the information and representations that would be required to be set forth in a shareholder's notice nomination for the election of directors pursuant to Section 2.11(a);
(B) the details of any relationship that existed within the past three years and that would have been described pursuant to Item 6(e) of Schedule 14N (or any successor item) if it existed on the date of submission of the Schedule 14N:

of a

(C) a representation and warranty that the shares of common stock of the corporation owned by the Nominating Shareholder were acquired in the ordinary course of business and not with the intent or objective to influence or change control of the corporation and are not being held with the purpose or effect of changing control of the corporation or to gain a number of seats on the Board of Directors that exceeds the maximum number of nominees that shareholders may nominate pursuant to this Section 2.15;

(D) a representation and warranty that the Nominating Shareholder satisfies the eligibility requirements set forth in Section 2.15(c) and has provided evidence of ownership to the extent required by Section 2.15(c)(i);

(E) a representation and warranty that the Nominating Shareholder will continue to satisfy the eligibility requirements described in Section 2.15(c) through the date of the Annual Meeting;

(F) a representation and warranty that the Nominating Shareholder has not nominated and will not nominate for election to the Board of Directors at the Annual Meeting any person other than the Nominees it is nominating pursuant to this Section 2.15;

(G) a representation and warranty that the Nominating Shareholder will not engage in, and will not be a "participant" in another person's, "solicitation" within the meaning of Rule 14a-1(l) (without reference to the exception in Section 14a-1(l)(2)(iv)) (or any successor rules) with respect to the Annual Meeting, other than with respect to its Nominees or any nominees of the Board of Directors;

(H) a representation and warranty that the Nominating Shareholder will not use any proxy card other than the corporation's proxy card in soliciting shareholders in connection with the election of a Nominee at the Annual Meeting;

(I) a representation and warranty that the Nominee's nomination for election to the Board of Directors or, if elected, Board membership would not violate applicable state or federal law or the rules of any stock exchange on which the corporation's securities are traded;

(J) a representation and warranty that the Nominee (1) qualifies as independent under the rules of any stock exchange on which the corporation's securities are traded, (2) meets the audit committee and compensation committee independence requirements under the rules of any stock exchange on which the corporation's securities are traded, (3) is a "non-employee director" for the purposes of Rule 16b-3 under the Exchange Act (or any successor rule), (4) is an "outside director" for the purposes of Section 162(m) of the Internal Revenue Code (or any successor provision), (5) meets the director qualifications set forth in Section 3.02, and (6) is not and has not been subject to any event specified in Rule 506(d)(1) of Regulation D (or any successor rule) under the Securities Act of 1933 or Item 401(f) of Regulation S-K (or any successor rule) under the Exchange Act, without reference to whether the event is material to an evaluation of the ability or integrity of the Nominee:

submission of the Nomination Notice;
(L) if desired, a statement for inclusion in the proxy statement in support of the Nominee's election to the Board of Directors, provided that such statement shall not exceed 500 words and shall fully comply with Section 14 of the Exchange Act and the rules and regulations thereunder, including Rule 14a-9; and
(M) in the case of a nomination by a group, the designation by all group members of one group member for purposes of receiving communications, notices and inquiries from the corporation and that is authorized to act on behalf of all group members with respect to matters relating to the nomination, including withdrawal of the nomination.
(iii) An executed agreement, in a form deemed satisfactory by the Board of Directors or its designee, acting in good faith, pursuant to which the Nominating Shareholder (including each group member) agrees:
(A) to comply with all applicable laws, rules and regulations in connection with the nomination, solicitation and election;
(B) to file any written solicitation or other written communication with the corporation's shareholders relating to one or more of the corporation's directors or director nominees or any Nominee with the SEC, regardless of whether any such filing is required under rule or regulation or whether any exemption from filing is available for such materials under any rule or regulation;
(C) to assume all liability (jointly and severally by all group members in the case of a nomination by a group) stemming from any action, suit or proceeding concerning any actual or alleged legal or regulatory violation arising out of any communication by the Nominating Shareholder, its affiliates and associates or their respective agents and representatives with the corporation, its shareholders or any other person in connection with the nomination or election of directors, including without limitation the Nomination Notice, or out of the facts, statements or other information that the Nominating Shareholder or its Nominees provided to the corporation in connection with the inclusion of such Nominees in the corporation's proxy statement;
(D) to indemnify and hold harmless (jointly with all other group members, in the case of a group member) the corporation and each of its directors, officers and employees individually against any liability, loss, damages, expenses or other costs (including attorneys' fees) incurred in connection with any threatened or pending action, suit or proceeding, whether legal, administrative or investigative, against the corporation or any of its directors, officers or employees arising out of or relating to any nomination submitted by the Nominating Shareholder pursuant to this Section 2.15 or a failure or alleged failure of the Nominating Shareholder to comply with, or any breach or alleged breach of, its obligations, agreements or representations under this Section 2.15; and
(E) in the event that any information included in the Nomination Notice, or any other communication by the Nominating Shareholder (including with respect to any group member) with the corporation, its shareholders or any other person in connection with the nomination or election ceases to be true and accurate in all material respects or omits a material fact necessary to make the statements made not misleading or that the Nominating Shareholder (including any group member) has failed to continue to satisfy the eligibility requirements

is, any entity whose principal business is the sale of specialty trucks and truck bodies or access equipment), within the three years preceding the

details of any position of the Nominee as an officer or director of any competitor of the corporation (that

(K)

requirements described in Section 2.15(c), as the case may be.

described in Section 2.15(c), to promptly (and in any event within 48 hours of discovering such misstatement, omission or failure) notify the corporation and any other recipient of such communication of the misstatement or omission in such previously provided information and of the information that is required to correct the misstatement or omission and/or notify the corporation of the failure to continue to satisfy the eligibility

- (iv) An executed agreement, in a form deemed satisfactory by the Board of Directors or its designee, acting in good faith, by the Nominee:
- (A) to make such other acknowledgments, enter into such agreements and provide such other information as the Board of Directors requires of all directors, including promptly completing the corporation's director questionnaire;
- (B) that the Nominee has read and agrees, if elected as a director of the corporation, to sign and adhere to the corporation's corporate governance guidelines and codes of ethics and any other corporation policies and guidelines applicable to directors; and
- (C) that the Nominee is not and will not become a party to (1) any compensatory, payment, reimbursement, indemnification or other financial agreement, arrangement or understanding with any person or entity in connection with service or action as a director of the corporation that has not been disclosed to the corporation, (2) any agreement, arrangement or understanding with any person or entity as to how the Nominee would vote or act on any issue or question as a director (a "Voting Commitment") that has not been disclosed to the corporation or (3) any Voting Commitment that could limit or interfere with the Nominee's ability to comply, if elected as a director of the corporation, with his or her fiduciary duties under applicable law.

The information and documents required by this Section 2.15(d) shall be (i) provided with respect to and executed by each group member in the case of information applicable to group members and (ii) provided with respect to the persons specified in Instruction 1 to Items 6(c) and (d) of Schedule 14N (or any successor item) in the case of a Nominating Shareholder or group member that is an entity. The Nomination Notice shall be deemed submitted on the date on which all the information and documents referred to in this Section 2.15(d) (other than such information and documents contemplated to be provided after the date the Nomination Notice is provided) have been delivered to or, if sent by mail, received by the Secretary of the corporation.

(e) Exceptions.

- (i) Notwithstanding anything to the contrary contained in this Section 2.15, the corporation may omit from its proxy statement and any ballot or form of proxy any Nominee and any information concerning such Nominee (including a Nominating Shareholder's statement in support), and no vote on such Nominee will occur (notwithstanding that proxies in respect of such vote may have been received by the corporation), and the Nominating Shareholder may not, after the Final Nomination Date, cure in any way any defect preventing the nomination of the Nominee, if:
- (A) the corporation receives a notice pursuant to Section 2.11(a) that a shareholder intends to nominate a person for election to the Board of Directors at the Annual Meeting;
- (B) the Board of Directors or its designee, acting in good faith, determines that such Nominee's nomination or election to the Board of Directors would result in the corporation violating or failing to be in compliance with these by-laws, the corporation's articles of incorporation or any applicable law, rule or regulation to which the corporation is subject, including any rules or regulations of any stock exchange on which the corporation's securities are traded:
- (C) the Nominee was nominated for election to the Board of Directors pursuant to this Section 2.15 at one of the corporation's two preceding Annual Meetings and withdrew or became ineligible or unavailable for election at any such Annual Meeting; or
- (D) the Nominee has been, within the past three years, an officer or director of a competitor, as defined for purposes of Section 8 of the Clayton Antitrust Act of 1914, as amended.

- (ii) Notwithstanding anything to the contrary contained in this Section 2.15, the corporation may omit from its proxy statement, or may supplement or correct, any information, including all or any portion of the statement in support of the Nominee included in the Nomination Notice, if the Board of Directors or its designee, acting in good faith, determines that:
- (A) such information is not true in all material respects or omits a material statement necessary to make the statements made not misleading;
- (B) such information directly or indirectly impugns character, integrity or personal reputation of, or directly or indirectly makes charges concerning improper, illegal or immoral conduct or associations, without factual foundation, with respect to, any person; or
- (C) the inclusion of such information in the proxy statement would otherwise violate SEC rules or any other applicable law, rule or regulation.

ARTICLE III. BOARD OF DIRECTORS

- 3.01 General Powers and Number. The business and affairs of the corporation shall be managed by its Board of Directors. The number of directors of the corporation shall be fixed from time to time exclusively by the Board of Directors pursuant to a resolution adopted by the affirmative vote of a majority of the total number of directors that the corporation would have if there were no vacancies. The Board of Directors shall annually choose, from among them, a Chairman of the Board, who shall serve as such until a successor is elected. The Chairman of the Board shall be a director that the Board has determined to be independent in accordance with the listing standards of the New York Stock Exchange Listed Company Rules in effect from time to time who has not previously served as an executive officer of the Company. The Chairman of the Board shall perform such duties and shall have such authority as are specified in these by-laws and as the Board of Directors may from time to time assign to him or her.
- 3.02 Tenure and Qualifications. Each director shall hold office until the next Annual Meeting, and until his or her successor shall have been elected, or until his or her prior death, resignation or removal. A Director may be removed from office by affirmative vote of a majority of the outstanding shares entitled to vote for the election of such director, taken at a meeting called for that purpose. A director may resign at any time by filing his or her written resignation with the Secretary, which shall be effective when the notice is delivered unless the notice specifies a later date. Directors need not be residents of the State of Wisconsin or shareholders of the corporation. No one shall be eligible for election as a director nor shall any directors be eligible for re-election after attaining the age of seventy-two without the prior approval of the Governance Committee of the Board of Directors, or of such other committee of the Board of Directors then in effect performing a similar function with respect to nomination of directors.

3.03 <u>Certain Director Resignations.</u>

(a) In an uncontested election of directors, any nominee for director who receives a greater number of votes "withheld" from his or her election than votes "for" his or her election (a "Majority Withheld Vote") will promptly tender his or her resignation to the Chairman of the Board following certification of the shareholder vote. For purposes of this by-law, a contested election is one in which the Chairman of the Board determines that, as of the Determination Date, the number of persons properly nominated to serve as directors of the corporation exceeds the number of directors to be elected. The "Determination Date" is (i) the day after the meeting of the Board of Directors at which the nominees for director of the Board of Directors for such election are approved, when such meeting occurs after the last day on which a shareholder may propose the nomination of a director for election pursuant to the articles of incorporation or these by-laws, or (ii) the day after the last day on which a shareholder may propose the nomination of a director for election in such election pursuant to the articles of incorporation or these by-laws, when the last day for such a proposal occurs after the meeting of the Board of Directors at which the nominees for director of the Board of Directors for such election are approved, whichever of clause (i) or (ii) is applicable.

- (b) The Resignation Committee (as defined in Section 3.03(e)) will promptly consider the resignation submitted by a director receiving a Majority Withheld Vote, and the Resignation Committee will recommend to the Board of Directors whether to accept the tendered resignation or reject it. In considering whether to recommend that the Board of Directors accept or reject the tendered resignation, the Resignation Committee will consider all factors deemed relevant by the members of such committee, including, without limitation, any stated reasons why shareholders "withheld" votes for election from such director and the qualifications of the director whose resignation has been tendered.
- (c) The Board of Directors will act on the Resignation Committee's recommendation no later than 90 days following the date of the shareholders' meeting at which the election occurred. In considering such committee's recommendation, the Board of Directors will consider the factors considered by such committee and such additional information and factors the Board of Directors believes to be relevant. Following the Board of Directors' decision, the corporation will promptly publicly disclose in a Current Report on Form 8-K filed with or furnished to, as applicable, the SEC the Board of Directors' decision whether to accept the resignation as tendered, including a full explanation of the process by which the decision was reached and, if applicable, the reasons for rejecting the tendered resignation.
- (d) Any director who tenders a resignation pursuant to this Section 3.03 will not participate in the recommendation of the Resignation Committee regarding whether or not to accept that director's tendered resignation. Any director who tenders a resignation pursuant to this Section 3.03 will not participate in the Board of Directors' deliberation or vote regarding whether or not to accept that director's tendered resignation. If directors constituting a quorum of the Board of Directors did not receive a Majority Withheld Vote in the election that prompted one or more directors to tender the resignations upon which the Resignation Committee and the Board of Directors must then act, then no director who tendered a resignation pursuant to this Section 3.03 in connection with such election will participate in the Board of Directors' deliberation or vote (i) regarding whether or not to accept any tendered resignations related to such election or (ii) regarding matters relating to the Resignation Committee as it relates to such resignation.
- (e) The "<u>Resignation Committee</u>" shall be (i) the Governance Committee of the Board of Directors, or such other committee of the Board of Directors then in effect performing a similar function, if (A) less than a majority of the members of such committee received Majority Withheld Votes in the election that prompted one or more directors to tender the resignations upon which the Resignation Committee must then act or (B) all of the directors received Majority Withheld Votes in such election or (ii) a committee of the Board of Directors appointed by the Board of Directors and consisting only of directors who did not receive Majority Withheld Votes, if a majority of the members of the Governance Committee of the Board of Directors, or such other committee of the Board of Directors then in effect performing a similar function, but not all of the directors, received Majority Withheld Votes at such election.
 - (f) This by-law will be summarized or included in each proxy statement relating to an election of directors of the corporation.
- 3.04 <u>Regular Meetings</u>. The Board of Directors may provide, by resolution, the time and place, either within or without the State of Wisconsin, for the holding of regular meetings without other notice than such resolution.
- 3.05 Special Meetings. Special meetings of the Board of Directors may be called by or at the request of the Chairman of the Board, the Chief Executive Officer or the Secretary and shall be called by the Secretary upon the written request of a majority of the directors then in office. If such meeting shall be called upon the written request of a majority of the directors, the date of the meeting shall be within ten days of receipt by the Secretary or, in his absence by any Assistant Secretary, of their request, at a time determined by such officer. The Chairman of the Board, the Chief Executive Officer or the Secretary calling any special meeting of the Board of Directors, except as otherwise provided by by-law, may fix any place, either within or without the State of Wisconsin, as the place for holding any special meeting of the Board of Directors called by them, and if no other place is fixed, the place of meeting shall be the principal business office of the corporation in the State of Wisconsin.

- 3.06 Notice; Waiver. Notice of meetings of the Board of Directors (unless otherwise provided pursuant to Section 3.04) shall be given by written notice delivered personally or mailed or given by email, telegram or facsimile to each director at his or her business address or at such other address as such director shall have designated in writing filed with the Secretary, in each case not less five days if by mail and not less than forty-eight hours if by email, telegram, telephone, teletype, telegraph, facsimile or other form of wire or wireless communication, or personal delivery. If mailed, such notice shall be deemed to be delivered when deposited in the United States mail so addressed, with postage thereon prepaid. If notice be given by telegram or facsimile, such notice shall be deemed to be delivered when the telegram is delivered to the telegraph company. Whenever any notice whatever is required to be given to any director of the corporation under the articles of incorporation or these by-laws or any provision of law, a waiver thereof in writing, signed at any time, whether before or after the time of meeting, by the director entitled to such notice, shall be deemed equivalent to the giving of such notice. The attendance of a director at a meeting shall constitute a waiver of objects thereat to the transaction of any business because the meeting is not lawfully called or convened. Neither business to be transacted at, nor the purpose of, any regular or special meeting of the Board of Directors need to be specified in the notice or waiver of notice of such meeting.
- 3.07 Quorum. Except as otherwise provided by law or by the articles of incorporation or these by-laws, a majority of the number of directors as provided in Section 3.01 shall constitute a quorum for the transaction of business at any meeting of the Board of Directors, but a majority of the directors present (though less than such quorum) may adjourn the meeting from time to time without further notice.
- 3.08 <u>Manner of Acting</u>. The act of the majority of the directors present at a meeting at which a quorum is present shall be the act of the Board of Directors, unless the act of a greater number is required by law or by the articles of incorporation or these by-laws.
- 3.09 <u>Conduct of Meetings</u>. In the absence of the Chairman of the Board, the Chief Executive Officer shall call the meeting of the Board of Directors to order, and shall act as chairman of the meeting, and in their absence any director chosen by the Directors present shall call the meeting of the Board of Directors to order and shall act as chairman of the meeting. The Secretary shall act as Secretary of all meetings of the Board of Directors but, in the absence of the Secretary, the presiding officer may appoint any Assistant Secretary or any director or other person present to act as secretary of the meeting.
- 3.10 <u>Vacancies</u>. Any vacancy occurring in the Board of Directors, including a vacancy created by an increase in the number of directors, may be filled until the next succeeding annual election by the affirmative vote of a majority of the directors then in office, though less than a quorum of the Board of Directors; provided, that in case of a vacancy created by the removal of a director by vote of the shareholders, the shareholders shall have the right to fill such vacancy at the same meeting or any adjournment thereof; and provided further, that a vacancy filled by the Board of Directors shall be filled by the vote of the remaining director(s) elected by the voting group of shareholders which would be entitled to fill that vacancy at a meeting of the shareholders.
- 3.11 <u>Compensation</u>. The Board of Directors, by affirmative vote of a majority of the directors then in office, and irrespective of any personal interest of any of its members, may establish from time to time a reasonable compensation for directors of the corporation; provided that persons who are directors and also are officers or employees of the corporation eligible shall be ineligible to receive compensation as directors. By affirmative vote of a majority of such directors, and irrespective of any personal interest of any of them, the Board of Directors also may establish, from time to time, reasonable compensation for each of the officers of the corporation. The Board of Directors, from time to time, may delegate its authority under this by-law to an appropriate committee. The Board of Directors also shall have authority to provide for or to delegate authority to an appropriate committee to provide for reasonable pensions, disability or death benefits, and other benefits or payments to directors, officers and employees, and to their estates, families, dependents or beneficiaries on account of services rendered by such directors, officers and employees of the corporation.
- 3.12 Presumption of Assent. A director of the corporation who is present at a meeting of the Board of Directors or a committee thereof of which he or she is a member at which action on any corporate matter is taken shall be presumed to have assented to the action taken unless his or her dissent shall be entered in the minutes of the meeting or unless he or she shall file his or her written dissent to such action with the person acting as the secretary of the meeting before the adjournment thereof or shall forward such dissent by registered mail to the Secretary of the corporation immediately after the adjournment of the meeting. Such right to dissent shall not apply to a director who voted in favor of such action.

- 3.13 Committees. The Board of Directors by resolution adopted by the affirmative vote of a majority of them then in office may designate one or more committees from time to time. Each such committee shall consist of at least one director and shall have those of the powers of the Board of Directors as shall be granted to such committee. Each such committee may exercise its power at times when the Board of Directors is not in session, subject to these by-laws and the Wisconsin Business Corporation Law. The Board of Directors also at any time may elect one or more of its members as alternate members of any such committee. Any such alternate, upon request by the Chairman of the Board, or in his or her absence the Chief Executive Officer, or in his or her absence the Chairman of such committee, may take the place of any absent member or members of the committee at any of its meetings. Except as provided by these by-laws or by resolution of the Board of Directors, each such committee shall fix its own rules governing the conduct of its activities as the Board of Directors may request.
- 3.14 <u>Unanimous Consent Without Meeting.</u> Any action required or permitted by the articles of incorporation or by-laws or any provision of law to be taken by the Board of Directors at a meeting or by resolution may be taken without a meeting, if a consent in writing, setting forth the action so taken, shall be signed by all of the directors then in office.
- 3.15 Telephonic Meetings. Except as provided by this by-law, any action required or permitted by the articles of incorporation or by-laws or any provision of law to be taken by the Board of Directors at a meeting or by resolution may be taken by a quorum of the Board of Directors at a telephonic meeting or other meeting utilizing electronic communication, if all participating directors are informed that a meeting is taking place at which official business may be transacted; if all participating directors simultaneously may hear each other during the meeting; if each participating director immediately is able to send messages to all other participating directors; and if all communication during the meeting immediately is transmitted to each participating director.

ARTICLE IV. OFFICERS

- 4.01 Number. The principal officers of the corporation shall be a Chief Executive Officer, a President, any number of Vice Presidents, a Secretary and a Treasurer, each of whom shall be elected by the Board of Directors. Such other officers and assistant officers as may be deemed necessary may be elected or appointed by the Board of Directors, the Chief Executive Officer or the President. Any two or more offices may be held by the same person, except that the same person may not hold the offices of President and Secretary, President and Treasurer or President and Vice President.
- 4.02 <u>Election and Term of Office</u>. The officers of the corporation to be elected by the Board of Directors shall be elected annually. Each officer shall hold office until his or her successor shall have been duly elected or until his or her prior death, resignation or removal.
- 4.03 Resignation; Removal. Any officer may resign at any time by delivering written notice to an officer of the corporation. A resignation shall be effective when delivered unless the notice specifies a later date which is accepted by the corporation. Any officer or agent may be removed by the Board of Directors whenever in its judgment the best interest of the corporation will be served thereby. Any officer or assistant officer elected or appointed by the Chief Executive Officer or the President may be removed by the person then holding the title of the officer that appointed such officer or assistant officer whenever in its judgment the best interest of the corporation will be served thereby. However, such removal shall be without prejudice to the contract rights, if any, of the person so removed. Election or appointment shall not of itself create contract rights.
- 4.04 <u>Vacancies</u>. A vacancy in any principal office because of death, resignation, removal, disqualification or otherwise shall be filled by the Board of Directors for the unexpired portion of the term.
- Chief Executive Officer and President. Subject to the control of the Board of Directors, the Chief Executive Officer shall be responsible for the control and general management of all of the business and affairs of the corporation. In the absence of the Chairman of the Board, he or she may preside at all meetings of the shareholders and of the Board of Directors. He or she shall see that all resolutions and orders of the Board of Directors and its committees are carried into effect. The President (if the President is not then the Chief Executive Officer) shall perform all duties incident to the office of President and such other duties as may be prescribed by the Board of Directors, by the Executive Committee or by the Chief Executive Officer from time to time. In the absence of the Chief Executive Officer (if the President is not then the Chief Executive Officer) or in the event of his death, inability or refusal to act, or in the event for any reason it shall be impracticable for the Chief Executive Officer to act personally, the President shall perform the duties of the Chief Executive Officer, and when so acting, shall have all the powers of and be subject to all the restrictions upon the Chief Executive Officer. Both the Chief Executive Officer and the President (if the President is not then the Chief Executive Officer) shall have authority to sign, execute and acknowledge, on behalf of the corporation, all deeds, mortgages, bonds, stock certificates, contracts, leases, reports and all other documents or instruments necessary or proper to be executed in the course of the ordinary business of the corporation, or which shall be authorized by resolution of the Board of Directors. Except as otherwise provided by law or the Board of Directors, either of them also may authorize any Vice President or other officer or agent of the corporation to sign, execute and acknowledge such documents or instruments in his place and stead. Each shall have authority, subject to such rules as may be prescribed by the Board of Directors, to appoint such agents and employees of the corporation as he or she shall deem necessary, to prescribe their powers, duties and compensation, and to delegate authority to them. Such agents and employees shall hold office at the discretion of the Chief Executive Officer and the President.

- 4.06 The Vice Presidents. In the absence of the President or in the event of his death, inability or refusal to act, or in the event for any reason it shall be impracticable for the President to act personally, the Vice President (or in the event there be more than one Vice President, the Vice Presidents in the order designated by the Board of Directors, or in the absence of any designation, then in the order of their election) shall perform the duties of the President, and when so acting, shall have all the powers of and be subject to all the restrictions upon the President. The Board of Directors may designate a Vice President as the Chief Financial Officer, in which event he or she shall have responsibility for all financial matters which affect the corporation other than those expressly provided for the Treasurer. Any Vice President may sign, with the Secretary or Assistant Secretary, certificates for shares of the corporation; and shall perform such other duties and have such authority as from time to time may be delegated or assigned to him or her by the Chief Executive Officer, by the President or by the Board of Directors. The execution of any instrument of the corporation by any Vice President shall be conclusive evidence, as to third parties, of his or her authority to act in the stead of the President.
- 4.07 The Secretary. The Secretary shall: (a) keep the minutes of the meetings of the shareholders and of the Board of Directors in one or more books provided for that purpose; (b) see that all notices are duly given in accordance with the provisions of these bylaws or as required by law; (c) be custodian of the corporate records and of the seal of the corporation and see that the seal of the corporation is affixed to all documents the execution of which on behalf of the corporation under its seal is duly authorized; (d) keep or arrange for the keeping of a register of the post office addresses of each shareholder which shall be furnished to the Secretary by such shareholder; (e) sign with the Chief Executive Officer or the President, or a Vice President, certificates for shares of the corporation, the issuance of which shall have been authorized by resolution of the Board of Directors; (f) have general charge of the stock transfer books of the corporation; and (g) in general perform all duties incident to the office of Secretary and have such other duties and exercise such authority as from time to time may be delegated or assigned to him or her by the Chief Executive Officer, by the President or by the Board of Directors.
- 4.08 The Treasurer. The Treasurer shall: (a) have charge and custody of and be responsible for all funds and securities of the corporation; (b) receive and give receipts for money due and payables to the corporation from any source whatsoever, and deposit all such moneys in the name of the corporation in such banks, trust companies or other depositories as shall be selected in accordance with the provisions of Section 5.04; and (c) in general perform all of the duties incident to the office of Treasurer and have such other duties and exercise such other authority as from time to time may be delegated or assigned to him or her by the Chief Executive Officer, by the President or by the Board of Directors. If required by the Board of Directors, the Treasurer shall give a bond for the faithful discharge of his duties in such sum and with such surety or sureties as the Board of Directors shall determine.

- 4.09 <u>Assistant Secretaries and Assistant Treasurers</u>. There shall be such number of Assistant Secretaries and Assistant Treasurers as the Board of Directors, the Chief Executive Officer or the President may from time to time authorize. The Assistant Secretaries may sign with the Chief Executive Officer, the President or a Vice President certificates for shares of the corporation the issuance of which shall have been authorized by a resolution of the Board of Directors. The Assistant Treasurers shall respectively, if required by the Board of Directors, give bonds for the faithful discharge of their duties in such sums and with such sureties as the Board of Directors shall determine. The Assistant Secretaries and Assistant Treasurers, in general, shall perform such duties and have such authority as shall from time to time be delegated or assigned to them by the Secretary or the Treasurer, respectively, or by the Chief Executive Officer, the President or the Board of Directors.
- 4.10 Other Assistants and Acting Officers. The Board of Directors, the Chief Executive Officer and the President each shall have the power to appoint any person to act as Assistant to any officer, or as agent for the corporation in his stead, or to perform the duties of such officer whenever for any reason it is impracticable for such officer to act personally, and such assistant or acting officer or other agent so appointed shall have the power to perform all the duties of the office to which he or she is so appointed to be assistant, or as to which he or she is so appointed to act, except as such power may be otherwise defined or restricted by the Board of Directors, the Chief Executive Officer or the President.
- 4.11 <u>Salaries</u>. The salaries of the principal officers shall be fixed from time to time by the Board of Directors or by a duly authorized committee thereof, and no officer shall be prevented from receiving such salary by reason of the fact that he or she is also a director of the corporation.

ARTICLE V. CONTRACTS, LOANS, CHECKS AND DEPOSITS; SPECIAL CORPORATE ACTS

- 5.01 Contracts. The Board of Directors may authorize any officer or officers, agent or agents, to enter into any contract or execute or deliver any instrument in the name of and on behalf of the corporation and such authorization may be general or confined to specific instances. In the absence of other designation, all deeds, mortgages and instruments of assignment or pledge made by the corporation shall be executed in the name of the corporation by the Chief Executive Officer or the President, or, in the absence of the Chief Executive Officer and the President, by one of the Vice Presidents, and by the Secretary, an Assistant Secretary, the Treasurer, an Assistant Treasurer, or Controller. When necessary or required by law, the Secretary or an Assistant Secretary shall affix the corporate seal to all such instruments. When an instrument has been executed in the manner provided by this Section, no party or third person shall be required to inquire into the authority of the officers signing for the corporation so to act.
- 5.02 <u>Loans.</u> No indebtedness for borrowed money shall be contracted on behalf of the corporation and no evidences of such indebtedness shall be issued in its name unless authorized by or under the authority of a resolution of the Board of Directors. Such authorization may be general or confined to specific instances.
- 5.03 <u>Checks, Drafts, etc.</u> All checks, drafts or other orders for the payment of money, notes or other evidences of indebtedness issued in the name of the corporation, shall be signed by such officer or officers, agent or agents, of the corporation and in such manner as shall from time to time be determined by or under the authority of a resolution of the Board of Directors.
- 5.04 <u>Deposits</u>. All funds of the corporation not otherwise employed shall be deposited from time to time to the credit of the corporation in such banks, trust companies or other depositaries as may be selected by or under the authority of a resolution of the Board of Directors.
- 5.05 Voting of Securities Owned by this Corporation. Subject always to the specific directions of the Board of Directors, (a) any shares or other securities issued by any other corporation and owned or controlled by this corporation may be voted at any meeting of security holders of such other corporation by the Chief Executive Officer or the President or, in the absence of both of them, by any Vice President then present, and (b) whenever in the judgment of the Chief Executive Officer or the President or, in the absence of both of them, by any Vice President, it is desirable for this corporation to execute a proxy or written consent in respect to any share or other securities issued by any other corporation and owned by this corporation, such proxy or consent shall be executed in the name of this corporation by the Chief Executive Officer or the President or, in the absence of both of them, by any Vice President, without necessity of any authorization by the Board of Directors, affixation of corporate seal, or counter-signature or attestation by another officer. Any person or persons designated in the manner provided by this Section as the proxy or proxies of this corporation shall have full right, power and authority to vote the shares or other securities issued by such other corporation and owned by this corporation the same as such shares or other securities might be voted by this corporation.

ARTICLE VI. SHARES AND THEIR TRANSFER

- 6.01 Certificates for Shares. The corporation may issue any shares of the classes or series of capital stock of the corporation without certificates to the full extent that the Secretary or Assistant Secretary of the corporation determines that such issuance is appropriate and allowed by applicable law and rules of the New York Stock Exchange, any such determination to be conclusively evidenced by the delivery to the corporation's transfer agent and registrar by the Secretary or Assistant Secretary of the corporation of an instrument referring to this by-law and providing instructions of the Secretary or Assistant Secretary of the corporation to the transfer agent and registrar to issue any such shares without certificates in accordance with applicable law. In any event, the foregoing authorization does not affect shares already represented by certificates until the certificates are surrendered to the corporation. Certificates, if any, representing shares of the corporation shall be in such form, consistent with law, as shall be determined by the Board of Directors. All certificates shall be signed by the Chief Executive Officer or the President or a Vice President and by the Secretary or an Assistant Secretary. All certificates shall be numbered consecutively or otherwise identified. The name and address of each person to whom a certificate is issued, together with the number of shares represented by the certificate and the date of its issue, shall be entered on the stock transfer books of the corporation. The name and address of each person to whom a share is issued without a certificate, together with the number of shares so issued and the date of their issue, shall be entered on the stock transfer books of the corporation. All certificates surrendered to the corporation shall be canceled. No new certificate for previously issued shares shall be issued until the outstanding certificate(s) for the same share shall have been surrendered and canceled, except as provided by Section 6.06.
- 6.02 <u>Facsimile Signatures and Seal</u>. The seal of the corporation on any certificate for shares may be a facsimile. The signatures of the Chief Executive Officer or the President or Vice President and the Secretary or Assistant Secretary upon a certificate may be facsimiles if the certificate is manually signed on behalf of a transfer agent, or a registrar, other than the corporation itself or an employee of the corporation.
- 6.03 <u>Signature by Former Officers</u>. In case any officer, who has signed or whose facsimile signature has been placed upon any certificate for shares, shall have ceased to be such officer before such certificate is issued, it may be issued by the corporation with the same effect as if he or she were such officer at the date of its issue.
- 6.04 Transfer of Shares. Prior to due presentment of a certificate for shares for registration of transfer, and prior to compliance with the customary procedures for transferring shares issued without a certificate, the corporation may treat the registered owner of such shares as the person exclusively entitled to vote, to receive notifications and otherwise to have and exercise all the rights and power of an owner. Where a certificate for shares is presented to the corporation with a request to register for transfer, the corporation shall not be liable to the owner or any other person suffering loss as a result of such registration of transfer if (a) there were on or with the certificate the necessary endorsements and (b) the corporation had no duty to inquire into adverse claims or has discharged any such duty. The corporation may require reasonable assurance that said endorsements are genuine and effective and in compliance with such other regulations as may be prescribed by or under the authority of the Board of Directors. Where the corporation receives a request to register for transfer shares issued without a certificate, the corporation shall not be liable to the owner or any other person suffering loss as a result of such registration of transfer if (a) the party requesting the transfer has complied with customary procedures for transferring shares issued without a certificate and (b) the corporation had no duty to inquire into adverse claims or has discharged any such duty.
- 6.05 <u>Restrictions on Transfer.</u> The face or reverse side of each certificate representing shares, or the written statement provided to shareholders for shares issued without a certificate, shall bear a conspicuous notation of any restriction imposed by the corporation upon the transfer of such shares.

- 6.06 Lost, Destroyed or Stolen Certificates. Where the owner of a share represented by a certificate claims that his certificate has been lost, destroyed or wrongfully taken, a new certificate shall be issued in place thereof, or the corporation shall enter the name and address of such owner on the stock transfer books of the corporation as a person to whom a share has been issued without a certificate, if the owner (a) so requests before the corporation has notice that such shares have been acquired by a bona fide purchaser, and (b) files with the corporation a sufficient indemnity bond, and (c) satisfied such other reasonable requirements as may be prescribed by or under the authority of the Board of Directors.
- 6.07 <u>Consideration for Shares</u>. The shares of the corporation may be issued for such consideration as shall be fixed from time to time by the Board of Directors, provided that any shares having a par value shall not be issued for a consideration less than the par value thereof. The consideration to be paid for shares may be paid in whole or in part, in money, on other property, tangible or intangible, or in labor or services actually performed for the corporation. When payment of the consideration for which shares are to be issued shall have been received by the corporation, such shares shall be deemed to be fully paid and nonassessable by the corporation. No certificate shall be issued for any share, and no entry on the stock transfer books of the corporation of a share to be issued without a certificate, until such share is fully paid.
- 6.08 Stock Regulations. The Board of Directors shall have the power and authority to make all such further rules and regulations not inconsistent with the statutes of the State of Wisconsin as it may deem expedient concerning the issue, transfer and registration of shares of the corporation, whether or not represented by certificates.

ARTICLE VII. INDEMNIFICATION

- 7.01 <u>Certain Definitions.</u> The following capitalized terms (including any plural forms thereof) used in this Article VII shall be defined as follows:
- (a) "Affiliate" shall include, without limitation, any corporation, partnership, limited liability company, joint venture, employee benefit plan, trust or other enterprise, whether domestic or foreign, that directly or indirectly through one or more intermediaries, controls or is controlled by, or is under common control with, the Corporation.
- (b) "<u>Authority</u>" shall mean the entity selected by the Director or Officer to determine his or her right to indemnification pursuant to Section 7.04.
- (c) "Board" shall mean the entire then elected and serving Board of Directors of the Corporation, including all members thereof who are Parties to the subject Proceeding or any related Proceeding.
- (d) "<u>Breach of Duty</u>" shall mean the Director or Officer breached or failed to perform his or her duties to the Corporation and his or her breach of or failure to perform those duties is determined, in accordance with Section 7.04, to constitute conduct as a result of which the Director or Officer is not entitled to mandatory indemnification under the Statute.
- (e) "Corporation," as used herein and as defined in the Statute and incorporated by reference into the definitions of certain other capitalized terms used herein, shall mean this corporation, including, without limitation, any successor corporation or entity to this corporation by way of merger, consolidation or acquisition of all or substantially all of the capital stock or assets of this corporation.
- (f) "<u>Director or Officer</u>" shall have the meaning set forth in the Statute; provided, that, for purposes of this Article VII, (i) "Director or Officer" shall include a director or officer of a Subsidiary (whether or not otherwise serving as a Director of Officer), (ii) the term "employee benefit plan" as used in the Statute shall include an employee benefit plan sponsored, maintained or contributed to by a Subsidiary and (iii) it shall be conclusively presumed that any Director or Officer serving as a director, officer, partner, member, trustee, member of any governing or decision-making committee, manager, employee or agent of an Affiliate shall be so serving at the request of the Corporation.

- (g) "<u>Disinterested Quorum</u>" shall mean a quorum of the Board who are not Parties to the subject Proceeding or any related Proceeding.
- (h) "Expenses" shall mean and include fees, costs, charges, disbursements, attorney fees and any other expenses incurred in connection with a Proceeding.
- (i) "<u>Liability</u>" shall mean and include the obligation to pay a judgment, settlement, penalty, assessment, forfeiture or fine, including an excise tax assessed with respect to an employee benefit plan, and reasonable Expenses.
- (j) "Party" shall have the meaning set forth in the Statute; provided, that, for purposes of this Article VII, the term "Party" shall also include any Director or Officer or employee of the Corporation who is or was a witness in a Proceeding at a time when he or she has not otherwise been formally named a Party thereto.
- (k) "Proceeding" shall have the meaning set forth in the Statute; provided, that, in accordance with the Statute and for purposes of this Article VII, the term "Proceeding" shall also include all Proceedings (i) brought under (in whole or in part) the Securities Act of 1933, as amended, the Securities Exchange Act of 1934, as amended, their respective state counterparts, and/or any rule or regulation promulgated under any of the foregoing; (ii) brought before an Authority or otherwise to enforce rights hereunder; (iii) any appeal from a Proceeding; and (iv) any Proceeding in which the Director or Officer is a plaintiff or petitioner because he or she is a Director or Officer; provided, however, that any such Proceeding under this subsection (iv) must be authorized by a majority vote of a Disinterested Quorum.
- (l) "<u>Statute</u>" shall mean the provisions of the Wisconsin Business Corporation Law, Chapter 180 of the Wisconsin Statutes, relating to indemnification and insurance for Directors, Officers and others, which are contained in Wisconsin Statutes Sections 180.0850 through 180.0859 as of September 20, 2010, as the same shall then be in effect, including any amendments thereto after September 20, 2010, but, in the case of any such amendment, only to the extent such amendment permits or requires the Corporation to provide broader indemnification rights than the Statute permitted or required the Corporation to provide prior to such amendment.
- (m) "Subsidiary" shall mean any direct or indirect subsidiary of the Corporation as determined for financial reporting purposes, whether domestic or foreign.
- 7.02 <u>Mandatory Indemnification of Directors and Officers</u>. To the fullest extent permitted or required by the Statute, the Corporation shall indemnify a Director or Officer against all Liabilities incurred by or on behalf of such Director or Officer in connection with a Proceeding in which the Director or Officer is a Party because he or she is a Director or Officer.

7.03 Procedural Requirements.

- (a) A Director or Officer who seeks indemnification under Section 7.02 shall make a written request therefor to the Corporation. Subject to Section 7.03(b), within sixty days of the Corporation's receipt of such request, the Corporation shall pay or reimburse the Director or Officer for the entire amount of Liabilities incurred by the Director or Officer in connection with the subject Proceeding (net of any Expenses previously advanced pursuant to Section 7.05).
- (b) No indemnification shall be required to be paid by the Corporation pursuant to Section 7.02 if, within such sixty-day period, (i) a Disinterested Quorum, by a majority vote thereof, determines that the Director or Officer requesting indemnification engaged in misconduct constituting a Breach of Duty or (ii) a Disinterested Quorum cannot be obtained.
- (c) In either case of nonpayment pursuant to Section 7.03(b), the Board shall immediately authorize by resolution that an Authority, as provided in Section 7.04, determine whether the Director's or Officer's conduct constituted a Breach of Duty and, therefore, whether indemnification should be denied hereunder.

(d) (i) If the Board does not authorize an Authority to determine the Director's or Officer's right to indemnification hereunder within such sixty-day period and/or (ii) if indemnification of the requested amount of Liabilities is paid by the Corporation, then it shall be conclusively presumed for all purposes that a Disinterested Quorum has affirmatively determined that the Director or Officer did not engage in misconduct constituting a Breach of Duty and, in the case of subsection (i) above (but not subsection (ii)), indemnification by the Corporation of the requested amount of Liabilities shall be paid to the Director or Officer immediately.

7.04 Determination of Indemnification.

- (a) If the Board authorizes an Authority to determine a Director's or Officer's right to indemnification pursuant to Section 7.03, then the Director or Officer requesting indemnification shall have the absolute discretionary authority to select one of the following as such Authority:
- (i) An independent legal counsel; provided, that such counsel shall be mutually selected by such Director or Officer and by a majority vote of a Disinterested Quorum or, if a Disinterested Quorum cannot be obtained, then by a majority vote of the Board;
- (ii) A panel of three arbitrators selected from the panels of arbitrators of the American Arbitration Association in Wisconsin; provided, that (A) one arbitrator shall be selected by such Director or Officer, the second arbitrator shall be selected by a majority vote of a Disinterested Quorum or, if a Disinterested Quorum cannot be obtained, then by a majority vote of the Board, and the third arbitrator shall be selected by the two previously selected arbitrators, and (B) in all other respects (other than this Article VII), such panel shall be governed by the American Arbitration Association's then existing Commercial Arbitration Rules; or
 - (iii) A court pursuant to and in accordance with the Statute.
- (b) In any such determination by the selected Authority there shall exist a rebuttable presumption that the Director's or Officer's conduct did not constitute a Breach of Duty and that indemnification against the requested amount of Liabilities is required. The burden of rebutting such a presumption by clear and convincing evidence shall be on the Corporation or such other party asserting that such indemnification should not be allowed.
- (c) The Authority shall make its determination within sixty days of being selected and shall submit a written opinion of its conclusion simultaneously to both the Corporation and the Director or Officer.
- (d) If the Authority determines that indemnification is required hereunder, the Corporation shall pay the entire requested amount of Liabilities (net of any Expenses previously advanced pursuant to Section 7.05), including interest thereon at a reasonable rate, as determined by the Authority, within ten days of receipt of the Authority's opinion; provided, that, if it is determined by the Authority that a Director or Officer is entitled to indemnification against Liabilities' incurred in connection with some claims, issues or matters, but not as to other claims, issues or matters, involved in the subject Proceeding, the Corporation shall be required to pay (as set forth above) only the amount of such requested Liabilities as the Authority shall deem appropriate in light of all of the circumstances of such Proceeding.
- (e) The determination by the Authority that indemnification is required hereunder shall be binding upon the Corporation regardless of any prior determination that the Director or Officer engaged in a Breach of Duty.
- (f) All Expenses incurred in the determination process under this Section 7.04 by either the Corporation or the Director or Officer, including, without limitation, all Expenses of the selected Authority, shall be paid by the Corporation.

7.05 Mandatory Allowance of Expenses.

- (a) The Corporation shall pay or reimburse from time to time or at any time, within ten days after the receipt of the Director's or Officer's written request therefor, the reasonable Expenses of the Director or Officer as such Expenses are incurred; provided, the following conditions are satisfied:
- (i) The Director or Officer furnishes to the Corporation an executed written certificate affirming his or her good faith belief that he or she has not engaged in misconduct which constitutes a Breach of Duty; and
- (ii) The Director or Officer furnishes to the Corporation an unsecured executed written agreement to repay any advances made under this Section 7.05 if it is ultimately determined by an Authority that he or she is not entitled to be indemnified by the Corporation for such Expenses pursuant to Section 7.04.
- (b) If the Director or Officer must repay any previously advanced Expenses pursuant to this Section 7.05, such Director or Officer shall not be required to pay interest on such amounts.

7.06 Indemnification and Allowance of Expenses of Certain Others.

- (a) The Board may, in its sole and absolute discretion as it deems appropriate, pursuant to a majority vote thereof, indemnify a director or officer of an Affiliate (who is not otherwise serving as a Director or Officer) against all Liabilities, and shall advance the reasonable Expenses, incurred by such director or officer in a Proceeding to the same extent hereunder as if such director or officer incurred such Liabilities because he or she was a Director or Officer, if such director or officer is a Party thereto because he or she is or was a director or officer of the Affiliate.
- (b) The Corporation shall indemnify an employee who is not a Director or Officer, to the extent he or she has been successful on the merits or otherwise in defense of a Proceeding, for all reasonable Expenses incurred in the Proceeding if the employee was a Party because he or she was an employee of the Corporation.
- (c) The Board may, in its sole and absolute discretion as it deems appropriate, pursuant to a majority vote thereof, indemnify (to the extent not otherwise provided in Section 7.06(b)) against Liabilities incurred by, and/or provide for the allowance of reasonable Expenses of, an employee or authorized agent of the Corporation acting within the scope of his or her duties as such and who is not otherwise a Director or Officer.
- 7.07 <u>Insurance</u>. The Corporation may purchase and maintain insurance on behalf of a Director or Officer or any individual who is or was an employee or authorized agent of the Corporation against any Liability asserted against or incurred by such individual in his or her capacity as such or arising from his or her status as such, regardless of whether the Corporation is required or permitted to indemnify against any such Liability under this Article VII.
- 7.08 Notice to the Corporation. A Director, Officer or employee shall promptly notify the Corporation in writing when he or she has actual knowledge of a Proceeding which may result in a claim of indemnification against Liabilities or allowance of Expenses hereunder, but the failure to do so shall not relieve the Corporation of any liability to the Director, Officer or employee hereunder unless the Corporation shall have been irreparably prejudiced by such failure (as determined, in the case of Directors or Officers only, by an Authority selected pursuant to Section 7.04(a)).
- 7.09 Severability. If any provision of this Article VII shall be deemed invalid or inoperative, or if a court of competent jurisdiction determines that any of the provisions of this Article VII contravene public policy, this Article VII shall be construed so that the remaining provisions shall not be affected, but shall remain in full force and effect, and any such provisions which are invalid or inoperative or which contravene public policy shall be deemed, without further action or deed by or on behalf of the Corporation, to be modified, amended and/or limited, but only to the extent necessary to render the same valid and enforceable; it being understood that it is the Corporation's intention to provide the Directors and Officers with the broadest possible protection against personal liability allowable under the Statute.

- 7.10 Nonexclusivity of Article VII. The rights of a Director, Officer or employee (or any other person) granted under this Article VII shall not be deemed exclusive of any other rights to indemnification against Liabilities or allowance of Expenses which the Director, Officer or employee (or such other person) may be entitled to under any written agreement, Board resolution, vote of shareholders of the Corporation or otherwise, including, without limitation, under the Statute. Nothing contained in this Article VII shall be deemed to limit the Corporation's obligations to indemnify against Liabilities or allow Expenses to a Director, Officer or employee under the Statute.
- 7.11 Contractual Nature of Article VII; Changes to Rights. This Article VII shall be deemed to be a contract between the Corporation and each Director, Officer and employee of the Corporation and any repeal or other limitation of this Article VII or any repeal or limitation of the Statute or any other applicable law shall not limit any rights of indemnification against Liabilities or allowance of Expenses then existing or arising out of events, acts or omissions occurring prior to such repeal or limitation, including, without limitation, the right to indemnification against Liabilities or allowance of Expenses for Proceedings commenced after such repeal or limitation to enforce this Article VII with regard to acts, omissions or events arising prior to such repeal or limitation. If the Statute is amended to permit or require the Corporation to provide broader indemnification rights than this Article VII permits or requires, then this Article VII shall be automatically amended and deemed to incorporate such broader indemnification rights.

ARTICLE VIII. AMENDMENTS

- 8.01 <u>By Shareholders.</u> These by-laws may be altered, amended or repealed and new by-laws may be adopted by the shareholders by affirmative vote of not less than a majority of the shares present or represented and entitled to vote thereon under the articles of incorporation at any Annual Meeting or Special Meeting at which a quorum is in attendance.
- 8.02 <u>By Directors</u>. These by-laws may also be altered, amended or repealed and new by-laws may be adopted by the Board of Directors by affirmative vote of a majority of the number of directors present at any meeting at which a quorum is in attendance; but no by-law adopted by the shareholders shall be amended or repealed by the Board of Directors if the by-law so adopted so provides.
- 8.03 <u>Implied Amendments</u>. Any action taken or authorized by the shareholders or by the Board of Directors, which would be inconsistent with the by-laws then in effect but is taken or authorized by affirmative vote of not less than the number of shares or the number of directors required to amend the by-laws so that the by-laws would be consistent with such action, shall be given the same effect as though the by-laws had been temporarily amended or suspended so far, but only so far, as is necessary to permit the specific action so taken or authorized.